



# SERIPLIS

### EMOTIONAL COMPETENCE HANDBOOK



### **Publication Information**

Title: Emotional Competence Handbook

Design: Project logo: Ritta Castro | Cover and back cover: Gabriel Cramariuc

Edition: First Edition

Date published: December 2023

URI: https://hdl.handle.net/1822/87373

**ISBN:** 978-989-35409-2-3

DOI: 10.21814/1822.87373

### COPYRIGHT AND CONDITIONS OF USE BY THIRD PARTIES

This is an academic work that may be utilized by third parties, provided internationally accepted rules and best practices regarding copyright and related rights are adhered to. The work may be used in accordance with the terms outlined in the license below. If the user requires permission to use the work under conditions not specified in the indicated license, please contact the author's through the University of Minho.

### License granted to the users of this work



Atribuição-NãoComercial-Sem Derivações CC BY-NC-ND

https://creativecommons.org/licenses/by-nc-nd/4.0/

### Acknowledgments:

The coordination team would like to express gratitude to Dr. Natalia Antunes, Dr. Dominique Correia, Dr. Ricardo Saraiva and Mrs. Guilhermina Pires for their unwavering support throughout the duration of this project.

### CONTENTS

		Page
PREFACE		7
EDITORS		8
AUTHORS		9
Chapter 1	Introduction	10
Chapter 2	Theories and Models of Emotional Intelligence	13
2.1	The theory of multiple intelligences	13
2.2	Emotional Intelligence as ability	14
2.3	Emotional intelligence as a trait	19
2.4	The Emotional Competences model	20
2.5	Mixed Models of Emotional Intelligence	24
Chapter 3	Emotional Regulation Interventions	31
3.1	Emotions in Healthcare	31
3.2	Emotion Regulation	32
3.3	Emotion Regulation in Healthcare	34
3.4	Intrapersonal and Interpersonal effects of emotion regulation in healthcare	37
3.5	Emotion regulation training for healthcare professionals	39
3.6	Emotion regulation interventions – Affect Regulation Training (ART)	41
Chapter 4	Problem-Based Learning in Nursing Education	52
4.1.	Introduction	52
4.2.	How to develop a PBL case-based study	54
4.3.	Positive/negative emotion case-based study	57
4.4.	How to evaluate the PBL with the students	58
Chapter 5	Emotional Competencies: Student Evaluation Approaches	62

5.1.	Introduction	62
5.2.	The oral examination: Triple jump approach	63
5.3.	Emotional Competencies Triple Jump case: Mary	65
5.4.	Assessment of Clinical Skills Using the Objective Structured Clinical Examination (OSCE)	69
5.5	Emotional Competencies OSCE examination scenario for role-play	71
Chapter 6	The Emotional Competence VEIGA Scale (ECVS)	75
6.1	Introduction	75
6.2	Emotional Competence	77
6.3	The Emotional Competence Dimensions-Operational Terms	79
6.4	Measuring Emotional Competence	84
Chapter 7	Virtual Reality in Emotional Competence Education	101
7.1	Virtual Reality	101
7.2	Virtual Reality in Nursing Education	102
7.3	The SG4NS Serious Game	104
Chapter 8	Conclusion and Future Recommendations	109
8.1	Summary	109
8.2	Future Recommendations for Nurse Educators' Contribution to Development of Emotional Competence Amongst Nursing Students	111

### PREFACE

Welcome to the Emotional Competence Course Handbook. This comprehensive guide is an integral component of our innovative pedagogical training program, meticulously designed to enhance your understanding and mastery of emotional competence. This program is one of the Intellectual Outputs of the Erasmus+ Project, "Serious Games - Developing Emotional Competence for Nursing Students."

In today's rapidly evolving world, emotional intelligence and competence are indispensable skills. They enable us to navigate the complexities of interpersonal relationships, communicate effectively, and achieve personal and professional success. Recognizing the crucial role that emotional competence plays in our lives, we have developed this course to provide you with a structured, engaging, and immersive learning experience.

The origins of this journey trace back to the hallowed halls of University of Minho when visionary professors from the School of Nursing identified a gap in nursing education. While technical proficiency was emphasized, emotional competence was often overlooked. In the complex and emotionally charged world of healthcare, these educators recognized that nurses required more than just clinical knowledge; they needed a profound understanding of emotional competence to provide holistic and compassionate care to patients.

This handbook serves as your compass throughout the course, offering clear and concise instructions, insightful explanations, and practical exercises. It is tailored to assist learners of all backgrounds, whether you are a student, a professional, or someone simply interested in furthering your understanding of emotions and their impact on your life.

We encourage you to embrace this learning experience with an open mind, a willingness to explore, and a commitment to your personal growth. Emotional competence is a lifelong endeavour, and this course will equip you with the knowledge and tools to navigate the intricate terrain of emotions.

As you embark on this transformative journey, remember that emotional competence is not just a skill; it is a way of life. We hope this course and handbook will empower you to master the art of understanding and managing emotions, both in yourself and in your interactions with others. By the end of this course, you will have a deeper appreciation of the profound impact emotional competence can have on your personal and professional life.

We extend our heartfelt gratitude to the entire team behind this project, whose dedication and expertise have made this training program and handbook possible. It's a testament to the power of collaboration and commitment, thanks to esteemed partners like the Polytechnic University of Bragança, the University of León, the University of Genoa, the University of Malta, and the Ştefan cel Mare University of Suceava. It symbolizes the harmonious fusion of minds, the convergence of expertise, and the shared vision to enhance nursing education.

It is our collective hope that nursing students will embrace the wisdom within and carry the torch of emotional competence into the healthcare settings they will soon grace.

Best wishes for a successful and transformative learning experience!

Coordinator of the Erasmus + Project SG4NS Prof. Dr. Lisa Gomes University of Minho, Portugal

### EDITORS

Lisa GOMES				
Annamaria BAGNASCO				
Daren CHIRCOP				
Gianluca CATANIA				
Giuseppe ALEO				
João CAINÉ				
Maria Augusta BRANCO				
Maria CASSAR				
Roderick BUGEJA				
Rui PEREIRA				

University of Minho, Portugal University of Genoa, Italy University of Malta, Malta University of Genoa, Italy University of Genoa, Italy University of Minho, Portugal Polytechnic Institute of Bragança, Portugal University of Malta, Malta University of Malta, Malta University of Minho, Portugal

### AUTHORS

University of Minho, Portugal	Lisa GOMES Filomena SOARES João CAINÉ Paulo NOVAIS Rui PEREIRA
Polytechnic Institute of Bragança, Portugal	Maria Augusta BRANCO
Polytechnic Institute of Cávado and Ave, Portugal	Vitor CARVALHO Eva OLIVEIRA Inés OLIVEIRA
Stefan cel Mare University of Suceava, Romania	Adina COLOMEISCHI Andrea URSU Alina IONESCU-CORBU Florin TIBU Liliana BUJOR
University of Genoa, Italy	Annamaria BAGNASCO Gianluca CATANIA Giuseppe ALEO
University of Leon, Spain	Letícia VALDEÓN Inés CASADO Letícia BARRIONUEVO
University of Malta, Malta	Maria CASSAR Daren CHIRCOP Georgios YANNAKAKIS Marvin ZAMMIT Roderick BUGEJA Silvi RUSI

### CHAPTER 1

### Introduction

Authors: Maria CASSAR, Daren CHIRCOP, Roderick BUGEJA

Emotional Intelligence is a fundamental element in the exercise of the nursing profession. The nurse's ability to perceive and evaluate one's and others' emotions; the ability to generate feelings and emotions in order to facilitate cognitive activities; the ability to understand the information driven by affects; and the ability to make a good emotion (self) management, in order to promote emotional and intellectual growth and socially adaptive relations during interactions with patients is a "high-order nursing practice skill based on the notion that emotional intelligence is vital to practice" (Bulmer Smith, 2009 p 1629). Scientific evidence has positively correlated higher emotional intelligence to enhanced clinical decision-making, collegial relationships, clinical environment knowledge utilization and inter-professional relationships at multiple levels.

The overall goal of the project is to achieve higher standards of clinical practice through enhanced nurses' intrapersonal and social emotional competencies and abilities. This project integrates two major dimensions of today's active and dynamic teaching-learning process and transversal skills development. In a digital era, it is fundamental to bring new approaches to this process. Augmented reality, by way of a serious game interface, helps create simulated practice in a virtual learning environment, adding a new dimension to pedagogy. This emerging field of teaching and learning in the health sciences affords unique methods and conditions for instruction that are not easily replicated with more traditional educational modalities.

The specific objectives of the SG4NS project are to: -

- 1. Structure an educational program (course format) of emotional skills in undergraduate nursing students
- 2. Build a digital interface (SG prototype) to support the improvement of emotional skills in nursing students
- 3. Train nursing teachers to develop competencies and pedagogical knowledge in the field of emotional intelligence / emotional competencies
- 4. Improve nursing students' personal and social skills related to Emotional Competence

The expected learning outcomes resultant from the *personal* emotional competencies and skills development component of the learning program are for nursing students to: -

- a. Demonstrate the ability to recognize and understand their own emotions and motivations;
- b. Demonstrate the ability to recognize individual strengths, weaknesses and needs related to emotion recognition and expression;
- c. Develop a flexible and constructive regulation of one's own emotions.

The expected learning outcomes resultant from the **social** emotional competence and skills development component of the learning programme are for nursing students to: -

- a. Demonstrate the ability to analyse emotions, needs and concerns of others;
- b. Demonstrate the ability to recognise and discuss a patient's emotional signs;
- c. Demonstrate the ability to view things from the point of view of others;
- d. Build a good relationship and clear communication in order to influence and inspire others through the use of one's own and other people's emotions.

This handbook sets the content and guidance for the Emotional Competence Skills Development Programme for Nursing Students. The following chapter outlines and discusses theories and models of emotional intelligence, starting with Gardner's Theory of Multiple Intelligences, Mayer and Solevey's Emotional Intelligence as Ability Model, Petrides' Emotional Intelligence as a Trait model, Bisquerra and Escoda's Emotional Competences model, and Goleman and Bar-On's mixed models of emotional intelligence.

Chapter 3 defines and debates emotion regulation interventions, paying specific attention to emotions encountered within healthcare settings and among healthcare personnel. Emotion regulation in this chapter is reviewed in both the intrapersonal and interpersonal dimensions. It also addresses emotion regulation training among healthcare professionals and to this effect suggests the Affect Regulation Training (ART) model of effective emotion regulation with practical exercises.

Chapter 4 presents a discussion about teaching methodologies in higher education that can be adapted to emotional competence skills' development among nursing students, with a specific focus on the Problem-Based Learning (PBL) model. This learner-centred approach draws upon Dewey's experiential learning theory and Vygotsky's socio-cultural theory to stimulate and expand students' learning motivation to improve clinical reasoning skills and critical thinking. This chapter provides step-by-step guidance on how to develop a PBL case-based study addressing positive / negative emotions, and also how to evaluate the PBL approach with students. Chapter 5 discusses student evaluation approaches that address both the intellectual and the interpersonal communication dimensions, to assess learning in emotional competence development. Furthermore, these approaches also assess critical thinking. The chapter addresses two main approaches, the oral examination (the Triple Jump Approach) and a clinical approach, or the Objective Structured Clinical Examination (OSCE). To achieve a holistic approach to evaluation and assessment of emotional competence, as a global construct, these approaches can be triangulated by way of a data collection instrument instituted upon ordinal variables that measure emotional competence, namely the Emotional Competence Veiga Scale (ECVS).

Chapter 6 outlines a background and discusses a rationale as to how the ECVS likert-type temporal frequency scale or *formulaire* measures emotional competence by way of Daniel Goleman's five dimensions of emotional intelligence: self-awareness, emotion management, self-motivation, empathy and management of social emotions.

Chapter 7 outlines Virtual Reality and discusses its remit and benefits to training health care professionals, and particularly in nursing education. Nevertheless, it also warns against the risks and other challenges of this technology in education calling for preventive measures to be instituted. This Chapter also presents the Serious Game devised for the project, namely an Escape Room invoking a spectrum of positive and negative emotions in its users.

In conclusion, Chapter 8 completes this handbook with a résumé of the rationale and purpose of the SG4NS project, its activities and outputs, and proposes future recommendations based on retrospection and insights earned during the conduct of this innovative and highly exciting pedagogical project, taking the journey of developing and consolidating emotional competence skill acquisition in professional nurse training further.

### CHAPTER 2

### Theories and Models of Emotional Intelligence

Authors: Aurora Adina COLOMEISCHI, Alina IONESCU – CORBU, Florin TIBU

In recent years, there has been a growing emphasis on the concept of emotional intelligence, with a wide array of definitions and explanations. In the Encyclopedic Dictionary of Psychology (6), the term "emotional intelligence" is explained beginning with the identification of its distant origins (in Thorndike's concept of "social intelligence") and its proximate origins (in Gardner's concept of "personal intelligences" – intra- and interpersonal). It is defined as "a difference in the degree to which people access, process, and use emotional information." We intend to present the main theories that conceptualize the phenomenon of emotional intelligence, extracting the main ideas and understandings in order to facilitate deeper understanding of how these theoretical frames could figure out the practice within medical settings, afterwards.

### 2.1 THE THEORY OF MULTIPLE INTELLIGENCES - HOWARD GARDNER

Howard Gardner's Theory of Multiple Intelligences, developed in 1985 in his work "Frames of Mind," revolutionized the classical view of the existence of a single type of intelligence and offered a renewed sense of hope in human development and the possibilities for differentiated education. According to Gardner, intelligence is the ability to solve problems and create valued products within a particular cultural context. The standard definition of intelligence suggests that it is innate and fixed in quantity, but Gardner's vision is flexible and provocative, based on abilities shared by the entire population. He asserts that individuals possess varying proportions of diverse intelligences (at least eight types). These premises have significant educational implications as they provide all children with opportunities to access information according to their dominant intelligence type. Educators can assist students in using their unique combination of intelligence to succeed in school. Based on his studies, Gardner distinguishes the following types of intelligence: linguistic, logical-mathematical, musical, spatial, naturalistic, kinesthetic, interpersonal, and intrapersonal. Recent research has led Gardner to define a ninth type of intelligence: existential intelligence.

Of particular significance in the conceptualization of social and emotional development are the intra- and interpersonal intelligences.

Intrapersonal intelligence involves self-awareness, understanding one's strengths and weaknesses, effective goal planning, monitoring and controlling thoughts and emotions, and the ability to monitor one's interactions with others. In Gardner's view, intrapersonal intelligence entails the presence of an efficient self-model and the capacity to transfer and use this information in self-regulation.

Interpersonal intelligence means thinking about others and understanding them, empathetic ability, recognizing differences among people, and appreciating their ways of thinking. It involves effective interactions with individuals within the family and society.

These two types of intelligence, initially configured by Gardner under the label of personal intelligence, encompass elements relevant to social and emotional development, serving as foundations for emotional and social competence that ensure success in individual life pursuits.

### 2.2 EMOTIONAL INTELLIGENCE AS ABILITY – MAYER & SALOVEY MODEL

One of the first approaches to the concept of emotional intelligence is promoted by John D. Mayer and Peter Salovey (14). The authors suggest that the definition of emotional intelligence should bridge the cognitive and emotional aspects of the psychological life (starting from the premise that traditionally, intelligence describes the cognitive dimension of the psyche): "the ability to perceive emotions, to access and generate emotions to assist thought, to understand emotions and emotional information, and to regulate emotions to promote emotional and intellectual growth." Mayer and Salovey present two definitions of the concept: firstly, "the ability to monitor one's own and others' feelings and emotions, to discriminate among them, and to use this information to guide one's thinking and actions." Secondly, a revised definition of emotional intelligence involving: "the ability to perceive accurately, recognize, and perceive emotions; the ability to access and/or generate feelings when they facilitate thinking; the ability to understand emotions to promote emotional knowledge; the ability to regulate emotions to promote emotional knowledge; the ability to regulate emotions to promote emotional knowledge; the ability to regulate emotions to promote emotional knowledge; the ability to regulate emotions to promote emotional knowledge; the ability to regulate emotions to promote emotional and intellectual growth."

The authors conceptualize emotional intelligence in a four-dimensional model that evolves from basic psychological processes to more psychologically integrated processes, and within each dimension, emotional skills are arranged in the order of their development:

I. Perception, Evaluation, and Expression of Emotions:

1. The ability to identify one's own emotions (at a physical, emotional, and cognitive level).

2. The ability to identify emotions in others, in art, etc., through language, sound, appearance, and behaviour.

3. The ability to express emotions accurately and express needs related to those experiences.

4. The ability to discriminate between correct/incorrect, honest/dishonest expressions of feelings.

II. Emotional Facilitation of Thinking:

1. Emotions are apriori to thinking by directing attention to important information.

2. Emotions are vivid and available enough to be generated in support of judgment and memory of feelings.

3. The amplitude of emotional dispositions changes the individual's perspective from optimism to pessimism, encouraging consideration of multiple viewpoints.

4. Emotional states encourage differential approaches to specific problems, such as when happiness facilitates inductive reasoning and creativity.

III. Understanding and Analyzing Emotions; Engagement of Emotional Knowledge:

1. The ability to label emotions and recognize relationships between words and emotions, as well as relationships between emotions themselves, such as the relationship between sympathy and love.

2. The ability to interpret the meanings emotions convey regarding relationships, such as the sadness that often accompanies loss.

3. The ability to understand complex emotions: simultaneous feelings of love and hatred or mixtures such as the overwhelming feeling as a combination of fear and surprise.

4. The ability to recognize transitions between emotions.

IV. Reflective Regulation of Emotions to Promote Emotional and Intellectual Growth:

1. The ability to be open to both pleasant and unpleasant emotions.

2. The ability to engage or detach reflectively from an emotion based on the degree of information or utility it offers.

3. The ability to reflectively monitor emotions in relation to oneself and others.

4. The ability for personal and others' emotional management by enhancing negative emotions and increasing positive ones, without suppressing or exaggerating the information they may convey.

Regarding the acquisition of emotional intelligence, Mayer and Salovey argue that emotional skills begin in the family, with good parent-child interactions (parents help children identify and name their emotions, respect their feelings, and start connecting them to social situations). It is noted that the opportunities for learning emotional skills differ based on the emotional development of parents who can initiate emotional-cognitive learning processes. The authors consider it possible to acquire emotional skills through the standard curriculum (literature, art, music, theater, values lessons) and through the development of specific educational programs aimed at promoting emotional intelligence.

In the Handbook of Emotional Intelligence, John Mayer, Peter Salovey, and David Caruso (15) present three aspects of emotional intelligence: as Zeitgeist, as a personality trait, and as a mental ability. The authors identify the terms "emotional intelligence" and "emotional competence" as the most useful new terms designated by the American Dialect Society for 1995 (linked to the release of Daniel Goleman's best-selling work, "Emotional Intelligence"). The authors examine three meanings for the term emotional

intelligence: (1) Zeitgeist indicates a tendency of the time, a cultural tendency, the spirit of the times characterizing the moment – so the term can be explored from a cultural and political perspective; (2) emotional intelligence describes a group of personality traits considered important for success in life; (3) emotional intelligence presupposes a set of skills that ensure the processing of emotional information.

To support the first meaning – Zeitgeist (spirit of the times), the authors take a historical perspective and discuss the periods of stoicism (with strong anti-emotional consequences) in opposition to the Romantic era (which promoted emotional expression in art). They move on to the 1960s, marked by emotional expressiveness and "escape from the rigors of intellect" (15) and bring up the principles of humanistic psychology (G. Allport, A. Maslow, C. Rogers), according to which people should exercise self-determination, experience emotions directly, and grow emotionally. A trend of the time is seen in the contrast between the conflict and integration view of the relationship between thought and emotion. While emotion is often considered in conflict with reason, emotional intelligence, by contrast, can be interpreted as describing societal practices that integrate emotion and thought (this idea is reinforced by new research showing that emotional and cognitive systems at the brain level are more integrated than initially thought). From a cultural perspective, emotional intelligence is viewed as an equalizing variable, starting from another trend of the time, namely the conflict between recognizing differences among people and emphasizing their equality. Emotional intelligence was popularized as a response to the Bell Curve (which made distinctions between social classes in the USA based on intelligence) by showing that it can be stronger than general intelligence, and that emotional skills can be learned.

The presentation of the second meaning of emotional intelligence (as a group of personality traits) starts with the terms used in the description of the concept: motivation, emotion, cognition, consciousness (these are the four basic processes that make up the foundation of personality). Other aspects of personality are the self-model and the model of the world that must be built through learning, as well as the self-model in the world.

To illustrate this meaning, the authors propose the following scheme (15, p.100):

Level 3	Mental Traits Relevant to Self			General Traits			
Mental	E.g., self-esteem, self-awareness,			E.g., extroversion, verbal intelligence,			
Traits	personal intelligence, ego strength		conscientiousness, dogmatism, friendliness				
Level 2	Models of the Self		Models of the Self in the N		Moo	lodels of the World	
Mental	E.g., self-concept,		World	ל E.g		knowledge of	
Maps	ideal self, identity, life		E.g., roles, a	attachments, pro		nunciation, expert	
	story	identificatio		ns, conduct	kno	wledge	
		-	rules				
Nivel 1	Fundamental	Funda	amental	Fundamenta	al	Consciousness	
Mecanisme	Motivations	Emot	ions	Cognitive		E.g., awareness,	
mentale	E.g., the need to	E.g., joy, sadness,		Operations		attention, the flow of	
Level 1	eat, drink,	anger	r, fear,	E.g., learr	ning,	consciousness	
Mental	sleep, be with	relate	ed to	recall,			
Mechanisms	others, self-	psychophysiology		judgment,			
	defense			comparison			

To support this perspective, Mayer, Salovey, and Caruso argue that there are definitions of emotional intelligence that overlap with multiple aspects of personality, potentially encompassing the individual's entire mode of operation in the world.

The third perspective presents the concept as an intelligence that processes and benefits from emotions, composed of mental capacities, abilities, and skills. Emotional intelligence as an ability is described in a four-dimensional model: emotional perception and identification (involving recognition and acquisition of information from the emotional system); emotional facilitation of cognition (using emotions to enhance cognitive processing); emotional understanding (cognitive processing of emotions); emotional management (self-management and the management of others' emotions).

The authors provide a circular description of the four-dimensional model of emotional intelligence proposed by Mayer and Salovey in 1997 (15, p.108). The first branch of the model represents the ability to perceive and express feelings. Emotional perception involves recording, tracking, and deciphering

emotional messages as they are expressed in facial expressions, voice, tone, art objects, or other cultural artifacts. The second dimension refers to emotional facilitation and describes how emotions influence cognition. The third dimension involves understanding and reasoning based on emotions. The ability to understand emotions facilitates understanding fundamental truths about human nature and interpersonal relationships. The fourth dimension pertains to emotional management and involves coping skills in the face of different emotional dispositions, flexibility, adaptability, and understanding of emotional progress made in established relationships with others.

# 2.3 EMOTIONAL INTELLIGNECE AS A TRAIT : THE TRAIT MODEL (KONSTANTINOS VASILIS PETRIDES, 2001)

The trait model of emotional intelligence proposed by Petrides comprises a series of characteristics and dispositions situated at the lower levels of personality of which individuals might be aware. More specifically, Petrides presented emotional intelligence as an individual's self-perception of one's personality traits related to emotion or as the trait emotional self-efficacy (18, 19). The trait model argues that each individual may inherit characteristics that are helpful in some contexts, and less helpful in others, arguing that there is no recipe for an emotionally intelligent person that would thrive in all contexts of life, which resembles the personality characteristics of individuals.

For this model, the authors (19, 21) presented a series of components of trait emotional intelligence and how individuals who would gain high scores see themselves. All these dispositions are organized under four main facets represented by well-being, self-control, emotionality, and sociability, while two of the components (adaptability and self-motivation) were reported as independent. In the following, we will present these dispositions and how the authors defined them.



### Figure 1: Trait model of emotional intelligence by Petrides

Source: adapted from Petrides (17)

Adaptability: individuals with high scores would see themselves as `flexible and willing to adapt to new conditions`, while lower scores would be obtained by individuals who have some reservations towards changes in general.

**Assertiveness:** individuals with high scores would be `forthright, frank, and willing to stand up for their rights`, while lower scores are typical to individuals who fall into a passive attitude and hesitate to fight for their rights or express their wishes.

**Emotion perception (self and others):** individuals with high scores for this facet are `clear about their own and other people's feelings`, whilst lower scores are characteristic to individuals for whom it is not very clear what they and others feel and why.

**Emotion expression:** high scores individuals are `capable of communicating their feelings to others`, while lower scores show difficulties in verbalizing emotions.

**Emotion management (others):** individuals who gain high scores are `capable of influencing other people's feelings`, while lower scores might show difficulties in modulating and outfacing others` feelings.

**Emotion regulation:** individuals who have high scores are `capable of controlling their emotions` and adjusting their emotional state to the situation, while lower scores individuals have difficulties in managing strong emotions or feelings.

**Impulse control:** individuals high in impulse control tend to think before acting and are `reflective and less likely to give in to their urges`, while lower scores might show tendencies to engage in unhealthy behaviours or actions.

**Relationship skills:** individuals with high scores are ` capable of maintaining fulfilling personal relationships` by applying listening skills and maintaining healthy bonds, while lower scores individuals have difficulties in maintaining close relationships and might have inappropriate behaviours towards family and friends.

**Self-esteem:** individuals with high scores are `successful and self-confident` and show a healthy attitude towards themselves, while lower scores might show a low level of self-respect and trust in individuals` self-worth.

**Self-motivation:** high scores in this component show individuals who are `driven and unlikely to give up in the face of adversity`, whilst lower scores show a tendency to lower sense of achievement and a higher rate of abandonment in tasks.

**Social awareness:** individuals with high scores are `accomplished networkers with superior social skills` because they see themselves as good negotiators and confident in their endeavors, while lower scores might show restricted social abilities.

**Stress management:** in this case, individuals with high scores are capable of withstanding pressure and regulating stress` because they healthily manage their emotions, while low scores show difficulties in adaptively applying stress management strategies.

**Trait empathy**: high scores show that individuals are `capable of taking someone else's perspective` and seeing things from another point of view, while lower scores show difficulties in understanding other persons` perspectives and might maintain narrower opinions.

**Trait happiness:** this disposition shows individuals who are `cheerful and satisfied with their lives` and are, generally, content about the way they live, while lower scores are shown in individuals who have a negative perspective about their current life.

**Trait optimism:** individuals with high scores see themselves as `confident and likely to look on the bright side of life`, having a positive view over their future in general, while lower scores may show more negative attitudes towards one`s future, which may lead to missing opportunities.

The trait model proved to have applications in multiple domains across various samples comprised of child, adolescent, and adult populations (16, 21). For example, in child and adolescent populations, the previously mentioned components predicted prosocial and antisocial behaviour (7, 12), they also predicted more leadership and cooperation in other age groups and geographical regions (12, 13). In adult populations, the model was linked to a better commitment to the organization, reduced stress levels, and more perceived flexibility and control over the job (20).

Overall, the model pleads for the promotion of a trait view over the emotional intelligence components, rather than a set of skills or abilities, debating the importance of integrating the social-cognitive and self-concept approaches.

### 2.4 THE EMOTIONAL COMPETENCES MODEL (BISQUERRA ALZINA AND PEREZ ESCODA, 2007)

Bisquerra and Escoda (4) conceptualized emotional intelligence as the acquisition and development of knowledge, talents, skills, and attitudes that would contribute to recognizing, understanding, and managing one's emotional state. They present their model as a series of five blocks, each encompassing several capacities and abilities that can be learned. As shown in Figure 2, the main five blocks are emotional awareness, emotional regulation, emotional autonomy, social competence, and life skills for well-being. Each of these blocks includes certain abilities and attitudes that would lead to high scores on the respective block. In the following, we will present the components of each described block.



Figure 2: Competence model of emotional intelligence by Bisquerra and Escoda (4)

Source: adapted from Bisquerra & Escoda (4)

**Emotional awareness** refers to the capacity to identify and name one's emotions and the emotions of others in various contexts. It comprises other three aptitudes: (a) being aware of one's emotions – being able to identify, name and, in some cases, even be aware of emotions in future contexts; (b) being able to properly name emotions; and (c) understanding the emotions of others – being able to correctly perceive the emotions others feel and reveal them in an empathic way.

**Emotion regulation** stands for the ability to manage emotions by being aware of the relationship between affect, behaviour, and cognition. It also includes being able to elicit positive affect for oneself. This block includes five other skills, as follows: (a) being aware of the relationship between emotion, cognition and behaviour, and the role of cognition in regulating both emotion and behaviour; (b) emotional expression – the ability to accurately express emotions; (c) emotion regulation – the ability to manage emotions to pursue a series of objectives and staying motivated but also to prevent deeply negative emotional states; (d) abilities to confront negative emotions by using healthy strategies to decrease their intensity and impact; and (e) the ability to generate positive emotions for oneself – being able to relax and look for improving one`s quality of life.

**Emotional autonomy** is comprised of self and relationship elements and covers seven areas of skills, including self-esteem, self-motivation, positive attitudes towards oneself, responsibility towards oneself and others, emotional self-efficacy, the ability to critically assess social norms, and resilience.

**Social competence** mainly refers to obtaining and maintaining good relationships with others. This block covers basic social abilities (such as listening or greeting others), showing respect towards others, good communication skills, skills in expressive communication, the ability to effectively communicate one's emotions to others, prosocial behaviour and cooperation, the ability to be assertive, being able to prevent and solve conflicts, and the capacity to manage emotional situations effectively.

Life skills and well-being encompasses a positive attitude towards living that would lead to establishing healthy and responsible goals to create a meaningful and good life. Some of the main components incorporated in this block consist of creating adaptive goals, making decisions that are relevant in social, personal, and professional contexts, being able to ask for and receive help, being an active, responsible, and critical citizen, the ability to look for the subjective wellbeing, and being able to look for favorable experiences in social, personal, and professional contexts.

#### 2.5 MIXED MODELS OF EMOTIONAL INTELLIGENCE

Somewhat similarly to how things unfolded when the first theoretical perspectives on emotions that emerged in the field of psychology more than a century ago (namely, the James-Lange and Cannon-Bard theories) were later challenged by Schachter and Singer's (22) Two-Factor theory of emotion, so too mixed models of Emotional Intelligence (EI) came to be proposed in the past two or three decades as alternatives to then-thought more simplistic (i.e., either ability-based or trait-based) models of EI that had been put forward no long before. These mixed models have tried to reconcile other views by making use of scientific advancements in the fields of affective neuroscience and psychological testing methodologies. The most prominent EI researchers that introduced mixed models in conceptualizing EI were Reuven Bar-On and Daniel Goleman.

In the view of American psychologist and science journalist Daniel Goleman, EI represents a construct that relies the person's inner characteristics (that are both individual and social in nature) but is also viewed as a construct that can be developed and optimized through exposure to complex social situations and re-assessing them through observation of own behaviour and motives, reflection and reframing; Goleman, therefore, sees EI not only as a combination of personal traits and abilities, but also as a relational system that is equipped with some degree of plasticity.

Goleman's work contributed vastly to the establishment of EI as a validated type of intelligence. Before that, cognitive intelligence (together with its validated measurement index Intelligence Quotient, or IQ), was considered to be the only umbrella term that would comprise distinct dimensions Researchers Boyatzis & Sala (5) stated that, in order for a proposed intelligence concept to be considered distinct from other types of intelligence manifestations, the following requirements need to be met: (1) The proposed concept is an observable behaviour; (2) It is linked to underlying biological structures that are activated through characteristic behaviours; (3) Has implications for long-term outcomes related to life, school, career; (4) Is different enough from other personality / behavioural concepts; and (5) Measurement properties, such as convergent and discriminant validity, are satisfied. Both Goleman, and Bar-On (see further below) have made significant advancements in linking EI dimensions to evolutionary-based facets of emotions and observed behaviours, proposing scales and instruments to measure distinct dimensions of EI and conducting research that demonstrated the significance of EI components in predicting individual and social behaviours.

Goleman is being credited with re-thinking the conceptual nature of EI and providing more scientific background for the concept, thus helping with meeting the requirements stated above by Boyatzis and Sala. More specifically, his work on EI has flourished with our raised understanding of the brain circuitry that is activated by social stimuli. The rapid development of affective neuroscience enabled Goleman to study how brain structures that are important for the processing of emotional stimuli can communicate with each other in rendering the optimal neural state for the individual to become aware of inner as well as social cues and how to manage negative emotions more effectively. In Goleman's view, the overactivation of the amygdala elicited during conflictual or negative mood states can compromise the individual's ability to self-regulate, identify solutions and become attuned to social signals. On the other hand, due to circuitry through which the amygdala is connected to the prefrontal cortex (an area that promotes self-regulation by allowing the individual to inhibit impulses, find solutions to problems and plan ahead) the individual can regulate negative emotions and return to stress-free mental state that promotes emotional stability and allows the individual to initiate constructive social contact.

Goleman initially proposed his EI theory in the 1990's, and later refined his view during the years 2000'. In his seminal 1995 book (9), he identified five components that are essential to EI: self-awareness (knowing one's emotions, goals, strengths and weaknesses), self-regulation (being able to divert

impulses and regulate negative emotions), motivation (directing inner resources towards goal accomplishment), empathy (being considerate of other people) and social skills (being able to establish rapport and maintain social relationships). This series of characteristics represents a combination of traits and abilities that appear to be in connection to biological substrates that promote emotion regulation and social relationships.

In his 2002 book, Goleman refined his initial EI theory by reducing the number of components to four, each of these reflecting the personal *versus* social domain of action and whether it encompasses the process of recognizing *versus* regulating emotions (11). The model has since remained virtually unchanged and Goleman made use of it in many of his subsequent publications and conference presentations.

Figure 3. Goleman's four EI model components and their main characteristics. Adapted from Goleman et al., (11).

	RECOGNITION	REGULATION			
PERSONAL	SELF-AWARENESS	SELF-REGULATION			
	emotional self-awareness	emotional control			
	focused and clear about goals	• <u>inner conflict resolution</u>			
	• <u>self-confident</u>	• effective with self work			
SOCIAL	SOCIAL AWARENESS	RELATIONSHIP MANAGEMENT			
	<u>attunded to others</u>	• <u>teamwork</u>			
	• <u>empathic</u>	<ul> <li><u>negotiating conflicts</u></li> </ul>			
	• aware of own impact on others	• inspirational leadership			

In trying to identify applicative value for his views, Goleman used his theory to discuss implications for promoting a more effective leadership, in particular. Before EI was even conceptualized, leadership skills and professional success were largely considered to be attributes of cognitive (or rational) intelligence and had little to do with emotional dynamics within and between people. Both scientists and leaders across professions were unaware of the impact that so-called 'soft skills' could have on organizational success, and dealing with emotions in the workplace was probably considered both irrelevant and a waste of time and resources. However, an increasing number of organizations across domains have nowadays become more attuned to the proven benefits of employing emotionally competent leaders, in terms of both employees' work satisfaction and company cost effectiveness.

Within just a few years after Daniel Goleman first published his scientific perspective on EI in 1995, Israeli psychologist Reuven Bar-On starting writing extensively on his own view on the concept, together with proposing, for the first time, a self-report instrument that aimed at measuring EI, which he called The Emotional Quotient Inventory (EQ-i; 1). This was the first time when the Emotional Quotient concept was used, and it has since become just as popular as its sister concept of IQ.

Two were the major influences that contributed to Bar-On's theory: Darwin's work on emotions and Thorndike's Social Intelligence theory. Regarding the impact that Darwin's evolutionary work had on the scientist, it should be noted that Bar-On considered that emotions served a pivotal and clearly adaptive role in our complex social world, by helping us to navigate through goals and tasks that inevitably involved working with other people. According to the model that he developed, "emotionalsocial intelligence is a cross-section of interrelated emotional and social competencies, skills and facilitators that determine how effectively we understand and express our-selves, understand others and relate with them, and cope with daily demands and challenges." (2). Just like Daniel Goleman, Bar-On thought that a solid conceptualization of EI needed reliance on some kind of biological system that would promote both emotional processing and social affiliation behaviours and he found it in Darwin's view on the role of emotions in socialization. On the other hand, Thorndike's theory on social intelligence and its key role played in human performance also proved critical in shaping Bar-On's concept of emotional intelligence. Darwin's writings on the adaptive role of emotions in the social realm and Thorndike's theory on social intelligence exerted such great impact on Bar-On that many times throughout his writings he came to refer to the concept of 'emotional and social intelligence' instead of just 'emotional intelligence'.

In Bar-On's view, EI comprises the following five components: (1) recognizing and understanding emotions and to ability to express feelings; (2) understanding how other people feel and the ability to

connect with them; (3) being able to deal with your emotions; (4) the ability to adapt to changing scenarios and solve problems of a personal and interpersonal nature; and (5) being able to exert positive affect and find inner motivation. One can easily notice how emotional and social constructs are intertwined in Bar-On's conceptualization of EI, to the extent to which one could speculate that the whole purpose of experiencing emotions is, in the theorist's view, to serve the individual's social purposes.

In designing his EI instrument, the EQ-i, Bar-On targeted the components stated above, which resulted in a 133-item self-report measure that yields one global EI score and 5 composite scale scores (Intrapersonal, Interpersonal, Stress Management, Adaptability and General Mood) with 15 subscales. Psychometric studies conducted by the author and others showed high reliability and validity, while predictive studies showed links between EI measured with the instrument and social interaction abilities, physical and psychological health and school and work performance (3). Its EI scale scores also correlate well with those measured through other EI tools.

<u>Conclusions</u>. Generally, the Goleman and Bar-On mixed models of EI aimed at an increased understanding of the intricate nature of processes that are activated in individuals when dealing with emotions in complex social contexts. Compared to alternative views (according to which EI was proposed to comprise solely abilities or solely traits), the mixed models have succeeded to incorporate in a meaningful and science-grounded manner two main ideas: 1) that there are neurobiological substrates underlying the EI, represented by both cortical (e.g., frontal and temporal areas) and subcortical (e.g., the amygdala) structures and which are clearly involved in expression of emotions and facilitation of social contact; and 2) that, in view of the modern neuroscience approaches, the EI components bear some degree of flexibility in development and are susceptible to change under favorable circumstances. This plasticity characteristic is given further support by research showing the role of mirror neurons as a neurobiological basis for empathy and modelling emotional competencies in general (10).

Besides being anchored in neurobiological-based approaches, the Goleman and Bar-On mixed models of EI have been successful in reaching beyond their formulated theoretical views by contributing to the design of applicative measurement tools with proven psychometric properties and which have been used widely in clinical practices, schools and leadership organizations. These mixed models on EI have become not just validated theoretical perspectives, but also significant advancements in modern social sciences that have inspired generations of individuals and organizational structures.

#### References:

- Bar-On, R. (1997). The Emotional Quotient Inventory (EQ-i): A test of emotional intelligence. Toronto, Canada: Multi-Health Systems, Inc.
- Bar-On, R. (2006). The Bar-On model of emotional-social intelligence (ESI). *Psicothema, 18 Suppl.*, 13–25.
- 3. Bar-On, R., Handley, R., & Fund, S. (2006). The impact of emotional and social intelligence on performance. In Vanessa Druskat, Fabio Sala, and Gerald Mount (Eds.), *Linking emotional intelligence and performance at work: Current research evidence*. Mahwah, NJ: Lawrence Erlbaum, pp. 3-19.
- Bisquerra, R., & Pérez Escoda, N. (2007). Las competencias emocionales. Educación XX1, 10, 61-82.
- Boyatzis, R. E., & Sala, F. (2004). The Emotional Competence Inventory (ECI). In G. Geher (Ed.), *Measuring emotional intelligence: Common ground and controversy* (pp. 147–180). Nova Science Publishers.
- 6. Davey, G. (2005). Encyclopaedic Dictionary of Psychology. Routledge. ISBN: 978-0-340-81238-9
- Frederickson, N., Petrides, K. V., & Simmonds, E. (2012). Trait emotional intelligence as a predictor of socioemotional outcomes in early adolescence. *Personality and individual differences*, 52(3), 323-328.
- 8. Gardner, Howard. (2006). Inteligențe multiple. Noi orizonturi. București: Editura Sigma.
- 9. Goleman, D. (1995). Emotional Intelligence. Bantam Books.
- 10. Goleman, D., & Boyatzis, R. (2008). Social intelligence and the biology of leadership. *Harvard business review*, *86*(9), 74–136.
- 11. Goleman, D., Boyatzis, R., & McKee, A. (2002). *Primal leadership: Realizing the power of emotional intelligence*. Harvard Business School Press.
- Mavroveli, S., Petrides, K. V., Sangareau, Y., & Furnham, A. (2009). Exploring the relationships between trait emotional intelligence and objective socio-emotional outcomes in childhood. British Journal of Educational Psychology, 79, 259-272.
- 13. Mavroveli, S. & Sánchez-Ruiz, M. J. (2011). Trait emotional intelligence influences on academic achievement and school behaviour. British Journal of Educational Psychology, 81, 112-134.

- 14. Mayer, John D., şi Salovey, Peter. (1997). What is Emotional Inteligence? În *Emotional Development and Emotional Inteligence. Educational Implications*. New York: Basic Books.
- 15. Mayer, John D., Salovey, Peter şi Caruso, David. (2000). Emotional Inteligence as Zeitgeist, as Personality and as Mental Ability în Bar-On, Reuven şi Parker, James, D.A. (coord). *The Handbook of Emotional Inteligence. Theory, development, Assesment, and Application at Home, School and Workplace.* San Francisco: Jossey-Bass A wiley Company.
- Pérez-González, J. C., Saklofske, D. H., & Mavroveli, S. (2020). Editorial: Trait Emotional Intelligence: Foundations, Assessment, and Education. *Frontiers in psychology*, *11*, 608. <u>https://doi.org/10.3389/fpsyg.2020.00608</u>
- Petrides, K. V. (2009). Psychometric properties of the Trait Emotional Intelligence questionnaire (TEIQue). In C. Stough, D. H. Saklofske, & J. D. Parker (Eds.), Assessing emotional intelligence. Theory, research, and applications (pp. 85-101). Boston, MA: Springer.
- Petrides, K. V. (2011). Ability and trait emotional intelligence. In T. Chamorro-Premuzic, S. von Stumm, & A. Furnham (Eds.), *The Wiley-Blackwell handbook of individual differences* (pp. 656– 678). Wiley Blackwell.
- 19. Petrides, K. V., & Furnham, A. (2001). Trait emotional intelligence: Psychometric investigation with reference to established trait taxonomies. European Journal of Personality, 15, 425-448.
- 20. Petrides, K. V., & Furnham, A. (2006). The role of trait emotional intelligence in a gender-specific model of organizational variables. Journal of Applied Social Psychology, 36, 552-569.
- 21. Petrides, K. V., & Mavroveli, S. (2018). Theory and applications of trait emotional intelligence. *Psychology: The Journal of the Hellenic Psychological Society, 23*(1), 24–36.
- 22. Schachter, S., & Singer, J. (1962). Cognitive, social, and physiological determinants of emotional state. *Psychological review*, *69*(5), 379.

## CHAPTER 3 Emotion Regulation Interventions

Authors: Liliana BUJOR, Andreea URSU

#### 3.1 Emotions in healthcare

Various individuals engaged in the healthcare process, such as medical practitioners, nurses, patients, and family members, encounter a range of complex emotions that exert notable influences, both beneficial and detrimental, on clinical outcomes (1; 2; 3; 4). Experiencing strong negative emotions and the inadequate management of these emotions have been found to be linked to medical students and healthcare professionals' psychopathology such stress, depression, anxiety and burnout and challenging interactions with patients. Additionally, patients' perspectives also indicate that these difficulties might lead to unsatisfactory relationships with healthcare providers. Moreover, these factors have the potential to shape the nature of interpersonal communication, work satisfaction, and the level of faith that patients place in their physicians. The recognition and comprehension of one's own emotions can facilitate the selection of suitable strategies for regulating emotions in order to achieve a desired objective. For instance, encouraging a patient to openly express their distress can result in reduced levels of anxiety, heightened satisfaction with their healthcare provider, and improved attitudes towards adherence to treatment and increasing job satisfaction among healthcare providers.

#### 3.2 Emotion regulation

One of the most widely used definitions of emotion regulation was proposed by Gross (5), which pertains to the cognitive and behavioural strategies employed by individuals to manage their emotions. These strategies involve the deliberate selection of specific emotions, as well as the timing and manner in which they are experienced or expressed. Five distinct emotion regulation strategies were presented in The extended process of emotion regulation (5, 6). These strategies include situation selection, situation modification, attention deployment, cognitive change, and response modulation. *Situation selection involves* actively choosing a situation where desired emotions are more likely to occur. *Situation modification* refers to altering a situation in order to change its emotional impact. *Attention deployment involves* redirecting one's attention within a given situation. *Cognitive change* entails modifying appraisals of a situation after it has occurred, thereby altering its emotional significance. Finally, *response modulation* involves directly influencing and modifying physiological, emotional, or behavioural responses. Within this model, cognitive reappraisal and expressive suppression were the most used in research studies.

Interpersonal emotion regulation is a concept that refers to the strategies employed by individuals to achieve a desired emotional state within the framework of a real-time social interaction (7). On the other hand, co-regulation refers to a specific form of interpersonal emotion regulation characterized by the reciprocal influence of emotional experiences, expressive behaviours, and physiological responses between the partners, leading to emotional and physiological stability for both individuals in an interpersonal relationship (8).

Over time, several approaches have been proposed for the conceptualization and classification of emotion regulation. Augustine and Hemenover (9) conducted a comprehensive examination of emotion regulation strategies, identifying six overarching categories: behavioural, cognitive, engagement, avoidance, antecedent-focused, and response-focused. According to Gross and Jazaieri (10), emotion regulation can be categorized as either intrinsic or intrapersonal, which consists of regulating of one's own emotions, or extrinsic or interpersonal, which consists of regulating of others' emotions. Similarly, Niven et al. (11) proposed that emotion regulation can be classified as either extrinsic or intrinsic, with the desire of worsening or enhancing one's own emotions or the emotions of others individuals. More specifically, the concept of intrinsic affect-improving refers to the intentional efforts made by persons in order to enhance their own emotional mood, while extrinsic affect-improving refers to the actions of persons to increase the experience of positive emotions of others individuals. The concept of intrinsic affect-worsening refers to the intentional desires of one person to induce negative emotions within oneself, while extrinsic affect-worsening refers to the conscious efforts of eliciting negative emotions in others persons.

The regulation of emotions can be categorized as either adaptive or maladaptive (12). For instance, adaptive emotion regulation strategies include reappraisal and acceptance, whereas maladaptive strategies include rumination and suppression. These strategies can be employed to either down-regulate or up-regulate emotions. Furthermore, the model proposed by Zaki and Williams (7) delineates two distinct categories of processes, namely intrinsic processes and extrinsic processes, as well as response-dependent processes and response-independent processes. Thus, the model posits the existence of four distinct forms of interpersonal emotion regulation.

Another area of investigation focused on umbrella concepts that encompass this emotional process. The three components of emotional competence, as identified by Scherer (13), include emotion perception, emotion regulation, and emotion production. On the other hand, emotional intelligence, as proposed by Salovey and Mayer (14), encompasses the processes of appraising and expressing emotions, regulating emotions, and utilizing emotions in adaptive manners. According to Scherer (13), emotion regulation pertains to an individual's aptitude to monitor and manipulate their own emotions in alignment with socio-cultural norms, strategic intentions, and the desire to maintain a state of equilibrium. In contrast, Salovey and Mayer (14) provide a definition for the regulation of emotions which refers to the capacity and inclination to observe, assess, and modify one's own and others' emotional responses in order to achieve a certain objective (14).

#### 3.3 Emotion regulation in Healthcare

Over the past twenty years, a significant body of research has been conducted to understand the associations between distinct emotion regulation strategies and mental health in general or clinical populations. The literature indicates that adaptive emotion regulation strategies, including cognitive reappraisal, actively approaching, positive refocus, positive reappraisal, seeking social support and acceptance are negatively correlated with psychological problems such as depressive symptoms, anxiety, and stress (15; 16; 17; 18); 19). Additionally, the use of maladaptive strategies, including rumination, withdrawal, catastrophizing, and ignoring, is positively linked to adverse consequences such as high levels of anxiety, stress, depression, and insomnia (18; 19; 20; 21). However, less is known on how emotion regulation affects healthcare personal wellbeing and healthcare personal-patient interactions.

In order to effectively navigate the complex dynamics of patient consultations, healthcare professionals must possess the necessary aptitude for emotion regulation. This entails the ability to effectively regulate and express their own emotions, while also creating a supportive environment that encourages patients to freely express their own emotional states. Related to healthcare professional emotion regulation strategies, one interesting line of research concerns which strategies are used by healthcare personal when experiencing strong positive and negative emotions in the presence of the patients. In order to address those high-arousal negative emotions, medical professionals employed a combination of both cognitive and behavioural strategies. The most used strategies by physicians were breathing, keeping the emotion away/focusing on something else, talking/listening to the patient, breaking eye contact, reframing, withholding the emotion (1). In their study, focusing on something else consists of ignoring; keeping emotional distance from the situation; continuing the encounter as if nothing was happening; focusing on the (next) task, on the patient, on one's posture; thinking as a professional; thinking of a solution; mentally counting, talking/listening to the patient refers to speaking calmly, gradually, with openness; maintaining silence; empathy; understanding; letting the patient express himself; keeping eye contact, breaking eye contact consists to gaining time; organizing thoughts, reframing consists of thinking of/providing an optimistic, hopeful perspective; rationalizing; accepting, withholding the emotion refers to blocking the feelings; not crying. It is widely accepted that breathing, talking/listening, and reframing are adaptive strategies with positive effects on subjective experience, such as decreasing negative emotions, or mental health issues as stress, anxiety and depression (16; 17; 22; 23). In contrast, keeping the emotion away/focusing on something else, withholding the emotion are considered maladaptive emotion regulation with detrimental effects on mental health (17; 19). The same study aimed at investigating the healthcare personal reactions when experiencing strong negative
emotions. Touching the patient, performing medical procedures, withdrawing from the situation and proving support were the most frequently reactions. Touching the patient and providing support to the patient are considered adaptive interpersonal emotion regulation strategies (24) while performing medical procedures and withdrawing from the situation can be maladaptive strategies (19). The results of another qualitative study suggest that emotional suppression was one of the most frequently used strategy by medical students when feeling that their emotions were impeding their ability to effectively carry out tasks (25). However, although in general, suppression is considered a maladaptive emotion regulation (17), in medical culture could be considered an adaptive strategy with positive consequences on the short-term (26). Furthermore, medical students implemented an additional strategy, which involved engaging in a process of reframing the stressful circumstances by rationalizing negative conduct and directing attention towards the positive abilities of a physician. The students exhibited a tendency to diminish the emotional intensity of the incident through the process of reappraisal, wherein they reassessed the issue and placed greater focus on external factors that may have contributed to the occurrence of the negative behaviour.

Two more recent studies assessed the emotional life of emergency department nurses and hospice nurses. The results of the study where emergency department nurses' emotion regulation strategies were assessed showed that disengagement/detachment, suppression, compartmentalization, distraction, cognitive reappraisal, seeking social support and exercising were the most frequently among this group (26). These findings indicate that a significant number of nurses hold the belief that their strategies are effective in mitigating the detrimental impact of negative emotions on their psychological wellbeing, patients and nurse-patient relationship. Additionally, the results on hospice nurses showed that they use the most extrinsic-affect improving, followed by intrinsic-affect improving.

The results also suggested that they use less frequently strategies such as extrinsic-affect worsening and intrinsic-affect worsening (27). This implies that hospice nurses typically seek to enhance others emotions, particularly for their patients and others involved in the care process.

#### 3.4 Intrapersonal and interpersonal effects of emotion regulation in Healthcare

The manner in which healthcare professionals respond to and regulate their emotions in clinical settings can have significant impacts on both the healthcare professionals themselves and the patients they treat, ultimately influencing the nature of the clinical relationship in profound manners.

In a study aiming at identifying the associations between emotion regulation and life satisfaction, compassion fatigue, and job satisfaction among hospice nurses it was found that intrinsic-affect improving was positively associated with life satisfaction, intrinsic-affect worsening was positively associated with compassion fatigue while extrinsic affect-worsening was negatively associated with lower job satisfaction (27). In other words, when hospice nurses use intrinsic-affect improving to regulate their emotions, they reported a high level of life satisfaction, one indicator of psychological wellbeing, while when hospice nurses use intrinsic-affect worsening they report high level of compassion fatigue. Additionally, hospice nurses who make others feel worse, in this case could be patients, family members of patients or colleagues, have lower job satisfaction. Another study investigated the links between long-term care nursing staff cognitive emotion regulation strategies and burnout (28). It was found that greater use of maladaptive cognitive emotion regulations strategies, such as rumination and blaming others was associated with higher levels of emotional exhaustion. Furthermore, positive reappraisal, an

adaptive cognitive emotion regulation strategy, correlated negatively with depersonalization. Additionally, refocus on planning and positive reappraisal were significantly and positively correlated with personal accomplishment among nurses.

Another line of research concerns the links between health care personal emotion regulation and the healthcare personal-patient relationship. Previous studies showed that patients' perceptions of healthcare personal emotion regulation skills were associated with nonverbal immediacy in doctor– patient interactions, patient satisfaction and communication quality (29). In the second study presented in same paper, it was found that strong and positive associations between healthcare professionals' ownreported emotion reappraisal and patients' satisfaction and positive affect before encountering the healthcare professionals. The links between maladaptive emotion regulation strategies and patienthealthcare professional were also investigated. In a study with physicians as participants, it was found moderate correlations between maladaptive interpersonal emotion regulation strategies, such as withdrawal, imposing oneself and defending oneself, and the occurrence of immediate negative consequences within the physician-patient relationship, increased defensiveness, avoidance of the patient, loss of empathy and of trust in the patient, and relationship termination (1).

Taken together, these results suggests that the strategies used by healthcare professionals to regulate their own emotions have important effects not only on their psychological wellbeing, but also on patients' psychological wellbeing and their satisfaction towards the encounter with the healthcare professional.

#### 3.5 Emotion regulation training for healthcare professionals

Having in mind the high prevalence of mental health problems, such as anxiety, stress, depression, Posttraumatic Stress Disorders and burnout among healthcare professionals (30; 31), various methods for preventing psychopathological symptoms were used in medical settings. One of them is developing competences to manage the aroused emotions and the challenging situations occurred in hospitals. In the following, we will present the results of interventions aiming to developed emotion regulation abilities among healthcare providers. For example, a study aimed to test the effectiveness of an emotion regulation intervention to decrease the occupational stress among Intensive Care Unit and Critical Care Unit nurses (32). The intervention consisted in eight sessions of two hours held during three weeks. During the first and last sessions, participants were invited to fill out the scales, while during the other six sessions, participants were provided with information about differences between various kinds of emotions, short-term and long-term effects of emotions, interpersonal skills (communication, expression and solving conflict), how to expand attention, shift attention and stop mental rumination, how to use cognitive adaptive strategies and how to change behavioural and physiological outcomes of emotions. The results showed that the nurses who were randomly assigned to the intervention group reported a lower level of occupational stress after the intervention. Additionally, they also reported lower levels of occupational stress dimensions such as conflict with physicians, problems with peers, workload, uncertainty concerning treatment and problems related to patients and their families compared to the control group. The results of another recent intervention study with also intensive and critical care nurses as participants present similar patterns of results (33). The emotion regulation training consists of acquiring information related to the psychoeducation and developing abilities such as progressive muscle relaxation, nonjudgmental

awareness, acceptance and tolerance of emotional responses, modification of attention, cognitive reappraisal, problem solving, and interpersonal skills. It was found that the intervention group reported lower levels of burnout, depression and anxiety compared to the waiting list group (control group). Additionally, the intervention group reported higher post-intervention scores of four out of five cognitive adaptive emotion regulation strategies (acceptance, refocus on planning, positive refocusing, positive reappraisal) and lower post-intervention scores for one maladaptive cognitive emotion strategy, catastrophizing.

Researchers were also interested in testing other emotion regulation trainings such as Affect Regulation Training (34). In the last section of this chapter, we will provide a detailed description of this model and some recommendations on how it can be used to develop emotion regulation skills among nurses. In the following, we will provide the results of an intervention study where the effectiveness of this training was tested (35). It was found that participants from the intervention group reported higher post-intervention improvements in emotion regulation skills such as acceptance, tolerance and modification. Furthermore, the ART-group reported also post-intervention improvements in wellbeing and improvements in emotion skills and wellbeing over all three point-time assessments (pre-intervention, post-intervention and follow-up). In other words, the results of this study showed that ART is effective in developing emotion regulation skills and improving wellbeing and maintaining the wellbeing of elderly care workers over a period of six months.

Taken together, the results of these interventions showed that emotion regulation trainings are reliable methods for developing emotion regulation skills and enhancing wellbeing at work, lowering psychopathological symptoms among healthcare professionals.

#### 3.6 Emotional regulation interventions- The Affect Regulation Training (ART)

The ART Model of Effective Emotion Regulation (34; 36) is a model of effective emotion regulation that utilizes techniques from: cognitive-behavioural therapy, dialectical behavioural therapy, mindfulnessbased interventions, empathy trainings, emotion focused therapy, and problem-solving therapies. The model demonstrated the importance of emotion regulation skills in psychotherapeutic interventions on the mental health status of hospitalized patients. In a German hospital for various mental disorders, a group of 289 patients benefited from cognitive behavioural therapy and integrative emotional competence training. According to the measurements carried out, if before the intervention through training the results of the test showed significant differences between the clinical group and the control group on all scales (especially on: acceptance, resilience and understanding), after the interventions, in the case of patients who benefited from both the usual treatment and an emotion regulation training, significant improvements were found in the emotion regulation skills scores. Improved coping skills, acceptance, and tolerance of negative emotions are the strongest and most consistent predictors of treatment gain (36). The training interventions modifies the emotional register: experience, expression, regulation. The data of a group experiment on students highlighted the fact that, following three-hour training sessions divided into two parts, the effect of training on emotional regulation skills, happiness and quality of life is significant in the case of experimental group (compared to control group). Therefore, people who learn to adaptatively regulate their emotions will report a higher level of happiness (37).

In *ART* Model, adaptative emotion regulation is conceptualized as a situation-dependent interaction between seven emotion regulation skills (36):

- 1. The ability to be consciously aware of emotions
- 2. The ability to identify and correctly label emotions
- 3. The ability to identify what causes and maintains an emotion
- 4. The ability to actively modify emotions in an adaptative manner
- 5. The ability to accept and tolerate negative emotions
- 6. The ability to approach and confront situations
- 7. The ability to provide compassionate self-support in distressing situations

#### Practice ART Model

In this section, you are invited to have some practical experiences that empower in your emotional life.

## Application 1. *Reflective moment*

1. How do you recognize an emotion to yourself?

2. Could you label the emotions?

3. How do you recognize an emotional storm?

4. How are the automatic thoughts in an adaptative/ non-adaptative emotion regulation?

# Application 2. The Truth about emotions

It is significant to have a correct and coherent perception about our emotions. According with your emotional life, establish the truth of the follow affirmations:

- 1. Letting others know that I am feeling bad is weakness. T/ F
- 2. Negative feelings are bad and destructive. T/ F
- 3. Being emotional means being out of control. T/ F
- 4. Emotions can just happen for no reason. T/ F
- 5. All painful emotions are a result of a bad attitude. T/ F

If you gave three or more answers with True, you have a lot of Myths about emotions.

# Application 3. Exercise the ability to be consciously aware and to correctly label emotions- Joy

- Propose ten words that describe the emotion of Joy.
- Propose (three-five) events for feeling Joy.
- Identify (three-five) aftereffects of Joy.

## Example:

- Ten words that describe the emotion of joy: amusement, contentment, delight, enthusiasm,
   hope, happiness, jolliness, joviality, pleasure, satisfaction.
- The aftereffects of Joy: *being friendly to others, doing nice things for other people, having a positive outlook; seeing the bright side* (Linehan, 1993)

# Application 4. Joy and Euphoria

- *Propose (three-five) events for feeling Euphoria.*
- Identify (three-five) aftereffects of Euphoria.
- Identify principals' differences between Joy and Euphoria. What do you observe? How could you explain these differences?

# Application 5. Sadness

- Propose ten words that describe the emotion of sadness.
- *Propose (three-five) events for feeling sadness.*
- Identify (three-five) aftereffects of sadness.

## Example:

- Ten words that describe the emotion of sadness: melancholy, hurt, disappointment, pity, sorrow, unhappiness, misery, loneliness, displeasure, gloom
- The aftereffects of sadness: *feeling tired or low in energy, feeling empty, feeling as if nothing is pleasurable any more* (38)

## Application 6. Sadness and Despair

- *Propose (three-five) events for feeling Despair.*
- Identify (three-five) aftereffects of Despair.
- Identify principals' differences between Sadness and Despair. What do you observe? How could you explain these differences?

# Application 7. Anger

- Propose ten words that describe the emotion of anger.
- Propose (three-five) events for feeling anger.
- Identify (three-five) aftereffects of anger.

## Example:

- Ten words that describe the emotion of anger: dislike, envy, exasperation, frustration, fury,

irritation, resentment, agitation, hostility, agitation

- The aftereffects of anger: *feeling incoherent, feeling out of control, muscle tightening* (Linehan, 1993)

#### Application 8. Anger and Destructiveness

- *Propose (three-five) events for feeling Destructiveness.*
- Identify (three-five) aftereffects of Destructiveness.
- Identify principals' differences between Anger and Destructiveness. What do you observe? How could you explain these differences?

## Application 9. The ability to identify what causes and maintains an emotion

Remember a recent situation when you were *Sadness* and another situation when you were *Despair*. Complete the table below with:

What effectively happened? What thoughts came into your mind? How do you decided to behave? What

were the consequences?

Date & Time	What happened?	Thoughts	Feelings	Behaviour	Consequences
			Sadness		
			Despair		

What do you observe about two emotions, causes and consequences?

What are helping thoughts?

Do helping thoughts give you other feelings?

What else could you do? Would that lead to other consequences?

# Application 10. The ability to identify what causes and maintains an emotion

Remember a recent situation when you were *Anger* and another situation when you were *Destructiveness*. Complete the table below with:

What effectively happened? What thoughts came into your mind? How do you decided to behave? What were the consequences?

Date & Time	What happened?	Thoughts	Feelings	Behaviour	Consequences
			Anger		
			Destructiveness		

What do you observe about two emotions, causes and consequences?

What are helping thoughts?

Do helping thoughts give you other feelings?

What else could you do? Would that lead to other consequences?

## Application 11. The ability to accept and tolerate negative emotions

Think at a specific situation when you experienced a negative emotion in a *loosing situation*. If you just looked at your body, from top to bottom, where you could identify some tensions. If this tension would have *a shape, a color and a name* (write in the space below) ...

# Application 12. The ability to provide compassionate self-support in distressing situations

If your best friend would be in a distressing situation, what would be your response in term of:

- a verbal message
- a nonverbal response
- a reward

Transfer these responses in relation with you, in a distressing situation.

# Some conclusions:

- The interventions for a functional emotional regulation have registered, in the last decades, an increasing attention from both researchers and practitioners
- The impact of the emotional regulation style on different variables (general well-being, affective style, anxiety, depressive mood) represent a strong argument for the integration of emotional regulation trainings in different levels of interventions
- The current trend, at the level of psychotherapy interventions, is to integrate the concepts,

techniques and methods belonging to several schools to enhance the results

#### References

- da Silva, J. V., & Carvalho, I. (2016). Physicians experiencing intense emotions while seeing their patients: what happens?. *Permanente Journal*, 20(3) 31-37. <u>https://doi.org/10.7812/TPP/15-229</u>
- Mjaaland, T. A., Finset, A., Jensen, B. F., & Gulbrandsen, P. (2011). Physicians' responses to patients' expressions of negative emotions in hospital consultations: a video-based observational study. *Patient Education and Counseling*, 84(3), 332-337. <u>https://doi.org/10.1016/j.pec.2011.02.001</u>
- Shapiro, J. (2011). Perspective: does medical education promote professional alexithymia? A call for attending to the emotions of patients and self in medical training. *Academic medicine*, *86*(3), 326-332. <u>https://doi.org/10.1097/ACM.0b013e3182088833</u>
- Van Middendorp, H., Lumley, M. A., Moerbeek, M., Jacobs, J. W., Bijlsma, J. W., & Geenen, R. (2010). Effects of anger and anger regulation styles on pain in daily life of women with fibromyalgia: A diary study. European Journal of Pain, 14(2), 176-182. <u>https://doi.org/10.1016/j.ejpain.2009.03.007</u>
- 5. Gross, J. J. (1998). The emerging field of emotion regulation: An integrative review. *Review of general psychology*, 2(3), 271-299. <u>https://doi.org/10.1037/1089-2680.2.3.271</u>
- 6. Gross, J. J. (2015). Emotion regulation: Current status and future prospects. *Psychological inquiry*, *26*(1), 1-26. <u>https://doi.org/10.1080/1047840X.2014.940781</u>

- 7. Zaki, J., & Williams, W. C. (2013). Interpersonal emotion regulation. *Emotion*, *13*(5), 803-810. <u>https://doi.org/10.1037/a0033839</u>
- 8. Butler, E. A., & Randall, A. K. (2013). Emotional coregulation in close relationships. *Emotion Review*, *5*(2), 202-210. <u>https://doi.org/10.1177/1754073912451630</u>
- Augustine, A. A., & Hemenover, S. H. (2009). On the relative effectiveness of affect regulation strategies: A meta-analysis. *Cognition and Emotion*, 23(6), 1181-1220. <u>https://doi.org/10.1080/02699930802396556</u>
- Gross, J. J., & Jazaieri, H. (2014). Emotion, emotion regulation, and psychopathology: An affective science perspective. *Clinical psychological science*, 2(4), 387-401. <u>https://doi.org/10.1177/2167702614536164</u>
- Niven, K., Totterdell, P., Stride, C. B., & Holman, D. (2011). Emotion Regulation of Others and Self (EROS): The development and validation of a new individual difference measure. *Current Psychology*, *30*, 53-73. <u>https://doi.org/10.1007/s12144-011-9099-9</u>
- 12. Aldao, A., Nolen-Hoeksema, S., & Schweizer, S. (2010). Emotion-regulation strategies across psychopathology: A meta-analytic review. *Clinical psychology review*, *30*(2), 217-237. <u>https://doi.org/10.1016/j.cpr.2009.11.004</u>
- 13. Scherer, K.R. (2007). Componential emotion theory can inform models of emotional competence. In Matthews, Zeidner, &Roberts (Eds.), *The science of emotional intelligence: Knowns and unknowns* (pp. 101–126). Oxford University Press.
- 14. Salovey, P., & Mayer, J. D. (1990). Emotional intelligence. *Imagination, cognition and personality*, *9*(3), 185-211. <u>https://doi.org/10.2190/DUGG-P24E-52WK-6CD</u>
- Carvajal, B. P., Molina-Martínez, M. Á., Fernández-Fernández, V., Paniagua-Granados, T., Lasa-Aristu, A., & Luque-Reca, O. (2022). Psychometric properties of the Cognitive Emotion Regulation Questionnaire (CERQ) in Spanish older adults. *Aging & Mental Health*, *26*(2), 413-422. <u>https://doi.org/10.1080/13607863.2020.1870207</u>
- 16. Garnefski, N., & Kraaij, V. (2006). Relationships between cognitive emotion regulation strategies and depressive symptoms: A comparative study of five specific samples. *Personality and individual Differences*, 40, 659-669. <u>https://doi.org/10.1016/j.paid.2005.12.009</u>
- 17. Gross, J. J., & John, O. P. (2003). Individual differences in two emotion regulation processes: implications for affect, relationships, and well-being. *Journal of personality and social psychology*, *85*(2), 348-362. <u>https://doi.org/10.1037/0022-3514.85.2.348</u>
- Jurado, M. D. M. M., del Carmen Pérez-Fuentes, M., Fernández-Martínez, E., Martínez, Á. M., & Linares, J. J. G. (2021). Coping strategies in the Spanish population: the role in consequences of COVID-19 on mental health. *Frontiers in Psychiatry*, *12*, 606621, 1-12. <u>https://doi.org/10.3389/fpsyt.2021.606621</u>

- Kraaij, V., & Garnefski, N. (2019). The behavioural emotion regulation questionnaire: development, psychometric properties and relationships with emotional problems and the cognitive emotion regulation questionnaire. *Personality and Individual Differences*, 137, 56-61. <u>https://doi.org/10.1016/j.paid.2018.07.036</u>
- 20. Dubey, N., Podder, P., & Pandey, D. (2020). Knowledge of COVID-19 and its influence on Mindfulness, Cognitive Emotion Regulation and Psychological Flexibility in the Indian community. *Frontiers in Psychology*, *11*, 3031, 1-18. <u>https://doi.org/10.3389/fpsyg.2020.589365</u>
- Ursu, A., & Măirean, C. (2022). Cognitive Emotion Regulation Strategies as Mediators between Resilience and Stress during COVID-19 Pandemic. International Journal of Environmental Research and Public Health, 19(19), 12631, 1-12. <u>https://doi.org/10.3390/ijerph191912631</u>
- Carvajal, B. P., Molina-Martínez, M. Á., Fernández-Fernández, V., Paniagua-Granados, T., Lasa-Aristu, A., & Luque-Reca, O. (2020). Psychometric properties of the Cognitive Emotion Regulation Questionnaire (CERQ) in Spanish older adults. *Aging & Mental Health*, 1-10. <u>https://doi.org/10.1080/13607863.2020.1870207</u>
- 23. Jerath, R., Crawford, M. W., Barnes, V. A., & Harden, K. (2015). Self-regulation of breathing as a primary treatment for anxiety. *Applied psychophysiology and biofeedback*, 40(2), 107-115. <u>https://doi.org/10.1007/s10484-015-9279-8</u>
- 24. Swerdlow, B. A., & Johnson, S. L. (2022). The Interpersonal Regulation Interaction Scale (IRIS): A multistudy investigation of receivers' retrospective evaluations of interpersonal emotion regulation interactions. *Emotion*, 22(6), 1119-1136. <u>https://doi.org/10.1037/emo0000927</u>
- 25. Doulougeri, K., Panagopoulou, E., & Montgomery, A. (2016). (How) do medical students regulate their emotions?. *BMC medical education*, *16*, 1-10. <u>https://doi.org/10.1186/s12909-016-0832-9</u>
- 26. Isbell, L. M., Boudreaux, E. D., Chimowitz, H., Liu, G., Cyr, E., & Kimball, E. (2020). What do emergency department physicians and nurses feel? A qualitative study of emotions, triggers, regulation strategies, and effects on patient care. *BMJ quality & safety*, *29*(10), 1-21. <u>http://dx.doi.org/10.1136/bmjqs-2019-010179</u>
- Barnett, M. D., Cantu, C., & Clark, K. A. (2020). Multidimensional emotion regulation strategies among hospice nurses. *Death Studies*, 44(8), 463-468. <u>https://doi.org/10.1080/07481187.2019.1586790</u>
- Bamonti, P., Conti, E., Cavanagh, C., Gerolimatos, L., Gregg, J., Goulet, C., ... & Edelstein, B. (2019). Coping, cognitive emotion regulation, and burnout in long-term care nursing staff: A preliminary study. *Journal of Applied Gerontology*, *38*(1), 92-111. <u>https://doi.org/10.1177/0733464817716970</u>
- 29. Kafetsios, K., Anagnostopoulos, F., Lempesis, E., & Valindra, A. (2014). Doctors' emotion regulation and patient satisfaction: A social-functional perspective. *Health Communication, 29*(2), 205-214. <u>https://doi.org/10.1080/10410236.2012.738150</u>

- 30. Wu, H., Chi, T. S., Chen, L. I., Wang, L., & Jin, Y. P. (2010). Occupational stress among hospital nurses: cross-sectional survey. *Journal of advanced nursing*, *66*(3), 627-634. <u>https://doi.org/10.1111/j.1365-2648.2009.05203.x</u>
- Serrano-Ripoll, M. J., Meneses-Echavez, J. F., Ricci-Cabello, I., Fraile-Navarro, D., Fiol-deRoque, M. A., Pastor-Moreno, G., ... & Gonçalves-Bradley, D. C. (2020). Impact of viral epidemic outbreaks on mental health of healthcare workers: a rapid systematic review and metaanalysis. *Journal of affective disorders*, 277, 347-357. <u>https://doi.org/10.1016/j.jad.2020.08.034</u>
- Saedpanah, D., Salehi, S., & Moghaddam, L. F. (2016). The effect of emotion regulation training on occupational stress of critical care nurses. *Journal of clinical and diagnostic research: JCDR*, 10(12), 1-4. <u>https://doi.org/10.7860/JCDR/2016/23693.9042</u>
- Kharatzadeh, H., Alavi, M., Mohammadi, A., Visentin, D., & Cleary, M. (2020). Emotional regulation training for intensive and critical care nurses. *Nursing & Health Sciences*, 22(2), 445-453. <u>https://doi.org/10.1111/nhs.12679</u>
- 34. Berking, M., Whitley, B., Berking, M., & Whitley, B. (2014). The adaptive coping with emotions model (ACE model). *Affect Regulation Training: A Practitioners' Manual*, 19–29.
- 35. Buruck, G., Dörfel, D., Kugler, J., & Brom, S. S. (2016). Enhancing well-being at work: The role of emotion regulation skills as personal resources. *Journal of occupational health psychology*, *21*(4), 480-493. <u>https://doi.org/10.1037/ocp0000023</u>
- Berking, M., Wupperman, P., Reichardt, A., Pejic, T., Dippel, A., & Znoj, H. (2008). Emotionregulation skills as a treatment target in psychotherapy. *Behaviour Research and Therapy*, 46(11), 1230–1237. <u>https://doi.org/10.1016/j.brat.2008.08.005</u>
- 37. Tahared mahdavi haji, Mohammadkhani, S., Hahtami, M., (2011), The Effectiveness of life skills on happiness, quality of life and emotion regulation. *Procedia- Social and Behavioural Sciences* 30, 407-411. <u>https://doi.org/10.1016/j.sbspro.2011.10.080</u>
- 38. Linehan, M. (1993). *Skills Training Manual for Borderline Personality Disorder,* The Guilford Press.

# CHAPTER 4 Problem-Based Learning in Nursing Education

Authors: Inés CASADO, Leticia BARRIONUEVO, Letícia VALDEÓN

#### 4.1. Introduction

Teaching methodologies in higher education have been in a state of constant evolution over the last few decades. UNESCO's report on Higher Education in the 21st Century emphasizes the relevance of higher education in addressing the many challenges facing humanity in this century and suggests the need for a shift from assigning grades to developing competencies in order to strengthen the link between education and the world of work [1]. This orientation stems from a reflection on the need to train graduates to embark on their professional lives with a sufficiently broad range of competencies, encompassing the pillars for education -learning to know, learning to do, learning to be, and learning to live together- as outlined in the Report of the International Commission on Education for the 21st Century [2]. The Organization for Economic Co-operation and Development's Project on the Definition and Selection of Competencies (DeSeCo) identifies the need to reorient curricula and adapt the role of the teacher to help learners acquire competencies [3].

Nowadays, in the field of health sciences, it is becoming increasingly apparent that there is a need to modify and improve traditional teaching methods in order to enhance student motivation and knowledge acquisition [4]. Future professionals will have to deal with problems that call for innovative approaches and specific skills to tackle complex issues. They must be trained using strategies and techniques that specifically apply to this reality [5].

Problem-Based Learning (PBL) is founded upon two conceptual and theoretical arguments. The first of these is the work of the educational philosopher John Dewey, who stressed the importance of learning through experience [6]. According to Dewey, in this real-world experience, students are confronted with a problem that stimulates their thinking, they gather the information they need to find solutions to the problem, and use this to test their knowledge. On the other hand, PBL draws on the socio-cultural theory of Vygotsky, who emphasized the importance of students participating in cognitive learning communities

by exchanging and comparing ideas with others, actively engaging with each other to solve problems while the teacher directs their efforts [6,7].

The use of this methodology dates back to 1969, when McMaster University School of Medicine in Canada introduced it into its curriculum. The aim was to find new approaches that would improve the students' motivation and learning, as they were very dissatisfied with the medical education methods used up to that time [8].

The first class to graduate from the new McMaster University School of Medicine did so in 1972. At the same time, the University of Michigan's Human Medicine program introduced a problem-based course as part of its pre-clinical curriculum. It was also in the early 1970s that the Universities of Maastricht (The Netherlands) and Newcastle (Australia) established medical schools that incorporated PBL into their curricular framework. At the beginning of the 1980s, other schools that had continued to work with a conventional curricular framework began to develop similar plans structured around this new approach. The university leading this trend was the University of New Mexico in the United States. Soon after, other schools accepted the challenge of restructuring their entire curricula around PBL. The universities leading this endeavor were Hawaii, Harvard, and Sherbrooke (Canada) [9].

Barrows (1986) defines PBL as "a learning method based on the principle of using problems as a starting point for the acquisition and integration of new knowledge." He recognizes that the patient diagnostic process was based on a combination of hypothetico-deductive reasoning and multi-disciplinary expertise. A "traditional" classroom-based approach to teaching discipline-specific content (anatomy, neurology, pharmacology, psychology, and so on) did not provide students with a context to apply this in the clinical setting [10]. This traditional approach was further confounded by the rapidly evolving knowledge base of Science and Medicine, which was driving changes in both theory and practice [11].

PBL is a learner-centered teaching and curricular approach that prepares the learner to carry out research, integrate theory and practice, and apply knowledge and skills to develop viable solutions to a defined problem. Fundamental to the success of this approach is the selection of ill-structured (and often interdisciplinary) problems and a tutor who guides the learning process and provides a comprehensive report at the end of the learning experience [11].

Barrows set out the essential characteristics of PBL, reduced to bullet points below [12]:

- Students must have the responsibility for their own learning.
- The problem simulations used in problem-based learning must be ill-structured and allow for free inquiry.
- Learning should be integrated from a wide range of disciplines or subjects.
- Collaboration is essential.
- What students learn during their self-directed learning must be applied back to the problem with reanalysis and resolution.
- A closing analysis of what has been learned from work with the problem and a discussion of what concepts and principles have been learned are essential.
- Self and peer assessment should be carried out at the completion of each problem and at the end of every curricular unit.
- The activities carried out in problem-based learning must be those valued in the real world.
- Student examinations must measure student progress towards the goals of problem-based learning.
- Problem-based learning must be the pedagogical base in the curriculum and not part of a didactic curriculum.

The PBL methodology is perfectly suited to the development of higher education systems because it facilitates not only the acquisition of knowledge, but also other skills such as teamwork, communication skills, assuming responsibility for independent learning, collaborating in the search for information, critically evaluating information, active listening, and respecting the points of view of others [13].

The aim of PBL in nursing education is to "improve clinical reasoning skills through problem solving and critical thinking among students" [14].

# 4.2. How to Develop a PBL Case-Based Study

Working on a real problem in the classroom begins by identifying and defining the professional competencies to be acquired, distinguishing between the different areas of knowledge, as described in the Delors Report [2]:

- Learning to know (technical competence).
- Learning to do (methodological competence).
- Learning to be (participative competence).
- Learning to live together (personal competence).

For a better understanding of the design and facilitation issues around PBL, we need to analyze the essential characteristics of this learning environment. PBL starts with a problem. This problem is the stimulus for learning. Complex problems with multiple objectives, some of which are imprecise, often have no single correct answer; there may be several solutions or none at all, forcing students to consider alternatives, choose the most viable solution, and provide a reasoned argument to support their choice [15].

In PBL, the student leads the process by searching for information, then selecting and organizing it with a view to solving the problem at hand. The teacher becomes a tutor, guiding the student's learning, suggesting sources of information, and responding to their needs. The aims are to develop thinking skills, activate cognitive processes, and transfer methodologies in order to "learn how to learn." PBL is therefore based on the assumption that learning is an active, integrated, and constructive process that is influenced by both social and contextual factors [9].

The development of a PBL case-based study involves seven key steps [16]:

- Step 1: clarifying unfamiliar terms. Unclear terms and concepts in the problem description are clarified, so that every group member understands the information that is given.
- Step 2: problem definition. The problem is defined in the form of one or more questions. The group has to agree upon the phenomena that need to be explained.
- Step 3: brainstorm. The preexisting knowledge of group members is activated and determined.
   This process entails the generation of as many explanations, ideas, and hypotheses as possible.
   The ideas of all group members are collected, without critical analysis.
- Step 4: analysing the problem. Explanations and hypotheses of the group members are discussed in depth and are systematically analysed. Ideas from the brainstorm are ordered and related to each other.
- Step 5: formulating learning goals. Based on contradictions, obscurities, and ambiguities from the problem analysis, questions are formulated that form the foundation for the study activities of the group members. In short, it is determined what knowledge the group lacks, and learning goals are formulated on these topics.

- Step 6: self-study. In the self-study phase, group members search for relevant literature that can answer the questions in the learning goals. After studying this literature, group members prepare themselves for reporting their findings in the next tutorial meeting.
- Step 7: reporting. After reporting what sources group members have used in their self-study activities, a discussion of the learning goals takes place based on the studied literature. Group members try to synthesize what they have found in different sources.



Figure 1. The seven-step approach [15]

## 4.3. Positive/Negative Emotion Case-Based Studies

#### Positive Emotion Case:

Mr. John is 64 years old and has been hospitalized for six days in a rehabilitation/physiotherapy unit following a stroke.

John is alert and oriented, and has right hemiparesis.

He immediately started motor rehabilitation with good results.

However, he tells the nurses that he is afraid that he will not be able to regain his motor skills and autonomy. His mood is not the best, he wants his life, his job, and his feelings back.

In the last two days John has reported being able to do exercises that he could not do before.

Enthusiastic about the improvements, he tells the nurse on duty that the situation is clearly improving.

Observing the positive recovery, the professional suggests that Mr. John also does the exercises in the afternoon and tries to do something independently, under the supervision of staff and in complete safety.

John is happy, at last he can see that healing is close at hand AND SAYS: "CAN WE START NOW?"

- Identify the patient's relationship needs.
- Discuss nursing interventions for the person with relationship needs in the light of evidencebased nursing.
- Identify the positive emotions present in this scenario.
- Discuss how to maintain positive emotions in both the patient and the healthcare professional.
- Discuss how positive emotions can drive improved health care in light of evidence-based nursing.

#### Negative Emotion Case:

A 46-year-old woman grieving the loss of her husband after a long illness. She has a teenage son and daughter, and another daughter of primary school age.

She was diagnosed years ago with an anxiety-depressive condition, which has been well controlled up to now with pharmacological treatment and private psychotherapy.

After the death of her husband she contacts her primary healthcare nurse. Frequent home visits were made throughout the course of her husband's terminal illness, fostering a relationship of trust and cooperation with the whole family. Now, because of her emotional and personal situation, it is she who turns to the nurse for support.

She is unemployed with a primary school education and no specialized training. Her current financial situation is precarious. Her husband held the family together, not just financially, and his death has left the whole family vulnerable, adding stress to an already complicated situation. The widow's pension is not enough to cover all expenses. She has no other family and her relationship with her husband's family is not very good.

A home interview is conducted to try to gain a deeper understanding of her emotional state and to improve her ability to cope with her current situation:

- Define grief and negative emotion.
- Identify the stages of development of negative emotions in the grieving process.
- Identify the main negative emotions that are likely to occur in the clinical case presented.
- Discussion of key nursing interventions for the appropriate emotional management of grief.
- Discuss how to manage emotions appropriately in order to avoid the onset of pathological grief.

## 4.4. How to Evaluate PBL with Students

In PBL, the evaluation process is integrated into the learning process and the feedback loop must be continuous, so that it serves as a stimulus for the improvement and development of the process itself. The main purpose of evaluation is therefore to provide feedback on strengths and weaknesses and to identify possible areas for improvement.

If changes are made in the way learning and teaching take place, it will also be necessary to change the way learning is evaluated. The aim is to determine whether the student has acquired the necessary knowledge through autonomous and co-operative learning and whether they have also developed and honed the expected competencies through deep reflection and playing an active role in shaping the learning process [17]. The main benefit of students' involvement in evaluation is that they become aware

of their formative process, can appreciate the contribution of others, and can make decisions based on their own judgments.

Some forms of assessment used in the PBL process are briefly described below [18]:

- Written exam. These can be closed or open book exams. The questions should be designed in such a way that the skills are transferable to similar problems or topics.
- Practical exam. These are used to ensure that the students are able to apply the skills they have learned during the course.
- Concept mapping. Students demonstrate their knowledge and cognitive growth by creating logical relationships between concepts and representing them graphically.
- Peer evaluation (co-evaluation). The student is provided with guidance on assessment categories to help in the peer assessment process. This process also underlines the co-operative work environment of PBL.
- Self-evaluation. This gives the learner the opportunity to think carefully about what they know, what they don't know and what they need to learn in order to perform certain tasks.
- Tutor evaluation. This consists of giving the tutor feedback on their involvement with the group.
   This can be given by the group or by an external observer.
- Oral presentation. PBL gives students the opportunity to practice their communication skills. Oral presentations are a way of observing these skills.
- Written report. This allows students to practice their written communication skills.

Assessment in PBL should be structured so that learners are able to demonstrate their understanding of problems and their solutions in a contextually meaningful way [5,17].

In PBL, two types of evaluation play a fundamental role [17]: formative and summative. Formative evaluation aims to determine whether the objectives set have been achieved. It is performed at the end of each learning task and its purpose is to analyze the results obtained and identify, if necessary, where any learning difficulties are to be found. This is carried out throughout the entire process. Summative evaluation aims, by means of a final assessment, to determine a grade according to how far the proposed objectives have been met, taking into account the value judgments that have been made about the student throughout the learning process.

#### References

1. UNESCO. La Educación Superior en el siglo XXI. Visión y Acción. París: UNESCO; 1998.

2. Delors J. La educación encierra un tesoro: informe de la Unesco de la Comisión Internacional sobre la Educación para el siglo XXI. Madrid: Santillana; 1996.

3. OCDE. La definición y selección de competencias clave. Resumen ejecutivo. París: OCDE; 2005.

4. Calzadilla ME. Aprendizaje colaborativo y tecnologías de la información y la comunicación. Revista Iberoamericana de Educación 2002;29(11):1-10. https://doi.org/10.35362/rie2912868

5. Escribano A, del Valle A, (Eds.). El Aprendizaje Basado en Problemas. Una propuesta metodológica en Educación Superior. Madrid: Narcea S.A. Ediciones; 2023.

6. Dewey J. Experience and Education: New York: Touchstone; 1938.

7. Eggen P, Kauchak D. Estrategias docentes. Enseñanza de contenidos curriculares y desarrollo de habilidades de pensamiento. México: Fondo de Cultura Económica; 2015.

8. Spaulding W. The Undergraduate Medical Curriculum (1969 Model): McMaster University. Can Med Assoc J. 1969;100 (14):659-64.

9. Barrows HS. Problem-based learning in medicine and beyond: A brief overview. In L Wilkerson, W Gijselaers, (Eds.). Bringing problem-based learning to higher education: theory and practice. San Francisco: Jossey-Bass; 1996. p. 3-12.

10. Barrows HS. Practice-based learning: Problem-based learning applied to medical education. Springfield: Southern Illinois University School of Medicine; 1994.

11. Savery JR. Overview of Problem-based Learning: Definitions and Distinctions. Interdisciplinary Journal of Problem-Based Learning 2006;1(1). https://doi.org/10.7771/1541-5015.1002

12. Barrows HS. The essentials of problem-based learning. Journal of Dental Education 1998;62:630-3. https://doi.org/10.1002/j.0022-0337.1998.62.9.tb03223.x

13. Wood D. ABC of learning and teaching in medicine. Problem based learning. BMJ 2003;326:328-30. http://dx.doi.org/10.1136/bmj.326.7384.328 14. Wosinskic J, Belcherb AE, Dürrenbergera Y, Allina AC, Stormacqa C, Gerson L. Facilitating problembased learning among undergraduate nursing students: A qualitative systematic review. Nurse Educ Today. 2018;60:67–4. http://dx.doi.org/10.1016/j.nedt.2017.08.015.

15. Hmelo-Silver CE, Barrows HS. Goals and strategies of a problem-based learning facilitator. Interdisciplinary Journal of Problem-Based Learning 2006;1(1):21–39. http://dx.doi.org/10.7771/1541-5015.1004

16. Camp G, van het Kaar A, van der Molen H, Schmidt H. PBL: step by step a guide for students and tutors. Rotterdam: Institute of Psychology, Erasmus University Rotterdam; 2014.

17. Ríos D. Sentido, criterios y utilidades de la evaluación del aprendizaje basado en problemas. Educ Med Super. 2007;21(3). http://bit.ly/1hFpD0s

18. Vicerrectoría Académica, Instituto Tecnológico y de Estudios Superiores de Monterrey. El Aprendizaje Basado en Problemas como técnica didáctica. Monterrey: Dirección de Investigación y Desarrollo Educativo; 2004.

# CHAPTER 5 Emotional Competences: Student Evaluation Approaches

Authors: Annamaria BAGNASCO, Gianluca CATANIA, Giuseppe ALEO

## 5.1 Introduction

To evaluate means to make a judgement and assign a score (1). In the last years, evaluation has become more of a strategy to determine the quality and value of a project, an outcome, or a curriculum (2). Indeed, in the field of nursing education, there is an increasing number of evaluation methods and nursing educators should identify and use the correct tool based on the competencies to be assessed (3). The evaluation phase involves the students, of course, but also the nursing tutor who needs to clearly know the educational objectives to be evaluated (2).

Regarding the educational programme about emotional competencies (EC), the evaluation involves different areas of learning: intellectual and interpersonal communication. Moreover, considering the importance of the PBL methodology used to acquire EC, nursing students also develop critical thinking (4).

To evaluate these different areas in this context, two main approaches can be used which are the oral examination through the triple jump approach and the Assessment of clinical skills using the Objective Structured Clinical Examination (OSCE). A quantitative data collection instrument, namely the Emotional Competence Veiga Scale (ECVS) to measure emotional competence is also recommended. Chapter 6 below presents a detailed description of this data collection instrument.

# 5.2 The oral examination – triple jump approach

The oral examination through the "triple jump" approach is an individual examination divided into three steps or "jumps" (2).

- *First step or jump*: a short, incomplete clinical case is presented to the student (5). Nursing students have to read and analyse it to identify and request missing information. In this first jump, the tutor evaluates the capacity of nursing students to elaborate on the situation and the mechanisms of critical thinking (2). At the end of this jump, students have to identify the nursing problems of the case presented. This jump required about 30 minutes.
- Second step or jump: students independently research and study how to solve the identified nursing problems. In this second jump, students research solutions using their notes taken during the lessons, books, and searching databases or medical records (2; 5). This jump required about 2 hours.
- *Third step or jump*: in the last jump students have to provide a final case summary where they report some hypotheses of nursing problem solutions. In this jump, students have to discuss with the tutor the summary of the problems and the nursing management plan (5). Tutors have to give some feedback to the students and encourage them to evaluate themselves. This jump required about 30 minutes (2).

Tutors evaluate each jump differently and independently. In the first jump, tutors have to evaluate if the student identifies all of the nursing problems, and if the student requires enough information. In the last jump, instead, tutors have to evaluate if the student identifies and prioritises correctly targeted interventions inherent to the case presented (2). The triple jump approach used in the emotional competencies program is reported in Chapter 4.3.

All of the jumps are reported in the following table (Table 4.1):

Table 4.1 Triple jump exam

First jump: definition of nursing problems (about 30 minutes)

Tutor activities	Students activities				
<ol> <li>Defining the outcome and the activities to evaluate.</li> </ol>	<ol> <li>Reading the case study.</li> <li>Identifying the missing information.</li> </ol>				
<ol><li>Presenting the case study to the student.</li></ol>	<ol> <li>Formulating the problem and generating hypotheses.</li> </ol>				
<ol> <li>Providing the additional information required by the student.</li> </ol>	4. Identifying study topics.				
4. Fill in the evaluation grid					

Second jump: identifying how to solve nursing problems (about 2 hours)

Tutor activities	Student's activities			
No activities.	<ol> <li>Searching information to understand the problem and solve it.</li> </ol>			
	2. Synthesising information.			

Third step: final case summary (about 30 minutes)

Tutor activities	Student's activities			
<ol> <li>Fill in the evaluation grid.</li> <li>Assessing the student's performance achievements.</li> </ol>	<ol> <li>Formulating the final problem and management plan, arguing the results obtained from the research.</li> <li>Self-assessing the level of performance.</li> </ol>			

# 5.3 Emotional Competencies Triple Jump case: Mary.

# <u>The case</u>

You are the nurse who is working the night shift, you are called incessantly by Ms. Maria to perform actions that you could do on your own (can you give me a drink, can you cover me better...). Maria is vigilant, oriented, and self-sufficient.

At each call, you go to answer promptly, and Maria greets you with a sarcastic smile of derision as if she wanted to challenge you.

On the fifth call, you present yourself to her worriedly as you realise that something is wrong.

## First jump (30 minutes)

1. Ask the student to formulate hypotheses about the case: *What nursing problems do you highlight in the case you have just read*?

Expected hypotheses about the case:

- Continuous and insistent calls caused by the need for communication inherent in concern about the ominous prognosis;
- Non-verbal communication caused by fear of death;
- Challenging attitude due to the need to be reassured about palliative therapy/anxiety related to the palliative care pathway.
- 2. Case-related data to provide to the student if required.

The student requests additional data needed to better frame the case. Mark the data requested by the student with a cross:

- □ Anthropometric values: weight 48 kg, height 165 cm.
- □ Personal data: age 58 years.
- □ Lives with her husband, has two daughters and four young grandchildren.
- □ Diagnosis: stage 4 lung cancer with bone metastases.
- □ No brain metastases.
- □ Ongoing chemotherapy cycle.
- □ Poor prognosis, recommended palliative therapies by oncologist.
- Maria is aware of her prognosis. Maria has been advised by her oncologist to stop the chemotherapy cycle and start palliative therapy to have the best possible quality of life in relation to her condition.
- Maria is aware that palliative therapy is the best choice but is confused and insecure. She cannot accept that her illness has 'come so far'. She was going ahead with chemotherapy and felt better than at the beginning. Apart from the strong cough she had resumed gardening with her husband. She just felt more tired lately.
- She felt uncertain about the palliative treatment; she knew that the chemotherapy was helping her. She was sure she would come out of it. She had hope of surviving the cancer.
- $\hfill\square$  She is a fraid of being abandoned by the doctors.
- $\Box$  She is a Buddhist.
- 3. Evaluation of the first jump.

Tutors have to complete this part of the following table (4.2). This table will be put together with table 4.3.

Table 4.2 first jump evaluation

Student's Activities							Score		
Has t	the student id	entified	at least	two	expected hypo	theses inheren	t in th	ne case	/5 <sup>1</sup>
prese	ented?								/⊃⁻
Was	the additiona	al inform	nation r	eques	sted appropria	<b>te</b> to define t	he ide	entified	
hypo	theses?								/5²
						h		- 11	
one	hypothesis	with	help;	3:	self-evident	hypothesis;	5:	all	hypoth

<sup>2</sup>1: the quality of the data collected is limited; 3: inadequate data collected relate to no more than one of the hypotheses; 5: the data collected are appropriate.

# Second jump (2 hours)

<sup>1</sup>1:

In this jump, tutors have to ensure that the student has:

- Identified the problem;
- Planned nursing interventions;
- Identified topics for independent study.

If the student has not identified key topics, note this on the study agenda. In this case, tutors have to help students to identify the topics.

## Third jump (30 minutes)

In the last jump, tutors have to:

- Identifying sources for bibliographic research and the use of databases;
- Evaluating if students apply new knowledge to the solution of problems/hypotheses;
- Evaluate the exposition of the student in a concise manner, through interventions, of the structured response to the problems addressed.

1. Evaluation of the third jump.

Tutors have to complete this part of the following table (6.3). This table will be put together with table 4.2. The sum of the score obtained from the first table and this one, gives to the tutors the total score of the triple-jump exam.

## Table 4.3 first third evaluation

Student Activities	Score
Has the student identified at <b>least two care problems</b> inherent in the case presented?	/7 <sup>1</sup>
Did the student <b>identify targeted interventions</b> relevant to the case presented?	/10 <sup>2</sup>
Has the student identified the <b>priority of interventions</b> inherent in the presented case?	/3 <sup>3</sup>

<sup>1</sup>1: one problem that does not reflect the use of theoretical knowledge; 4: problems that are not consistent with the use of theoretical knowledge; 7: problems that are integrated with theoretical knowledge.

<sup>2</sup>1: interventions not relevant to the case; 5: unclear and unrelated interventions; 10: interventions are very
 clear and relevant to the case.
 <sup>3</sup>1: inadequate assessment of priorities; 2: sets priorities appropriately; 3: clear evaluation of priorities with explanations.

# Self-assessment of the students

Tutors have to encourage students to identify weak points and strengths. The results obtained must be shared with the student in order to provide him/her with feedback. Indeed tutors have to answer the following questions:

- Did the student ask for clarification?
- Did the student understand the information provided?
- Did the student hypothesise a research path?
- Did the student maintain an active attitude?

The answers to these questions provide an overall assessment of the student.

# 5.4 Assessment of clinical skills using the Objective Structured Clinical Examination (OSCE)

The Objective Structured Clinical Examination (OSCE) is widely used for the evaluation of clinical skill performance in particular focusing on communication skills (6). With OSCE methods, three domains are evaluated: cognitive, affective, and psychomotor (7). The use of an observational grid allows the equity of the assessment of the student's performance (8). Through the OSCE, communicative competencies are evaluated based on the performance of the students during a simulated practical situation (3). The simulation represents a lifelike clinical scenario where the student must put himself at stake using his skills (6). In the EC educational programme, the skills to be assessed are emotional and communicational competencies. Indeed, tutors with OSCE examination could also assess students' problem-solving skills (9). It is important that the OSCE examination is conducted in an equipped setting to assess all the skills reported and, in a centre, where the simulation could be audio-visual recorded (6).

Before conducting an OSCE examination, it is important to prepare a lifelike simulation. The simulation is based on the role-playing method (7). The role of the patient should be played out by a voluntary nurse or an actor. It is important that the students feel like a real-life case, so, it could be a good idea to ask them to wear pyjamas to the "fake patient". At the same time, the students have to wear the normal uniform used during the clinical placement.

The scenario has to focus on the competencies to be achieved by the students at the end of the educational programme. Therefore, in this case, the scenario created is focused on emotional competencies and communication skills. In the script, it is necessary to insert a section regarding learning objectives and the behaviours expected from the student (6). The central section of the script is the patient history and his/her health problems. The script presents a part with information that the "fake

patient" spontaneously gives to the students and some others that should be requested by the students. The concluding section of the script outlined the student's assignment, detailing the goals to accomplish, key information about the role of the students, the setting context, and the clinical details documented in the nursing file (9).

To evaluate the students' performance a validated assessment grid is needed. The assessment grid evaluates the communication of relevant information, the use of appropriate language, active listening, and the clarity of communication. The assessment grid is composed of four constructs: terminology, listening, attention, and clarity. Each construct should be evaluated between -2 and +2 scores (table 4.4) (8).

	-2	-1	0	+1	+2
Terminology	Too detailed not appropriate		Appropriate. Not very precise	does not answer questions	Answers questions Immediately
Listening	Hears but does not listen	does not re-	Listens and re- states but not always correctly	Listens and re- states correctly	Check whether the fake patient has understood
Attention	Pays no verbal or behavioural attention		Pay verbal attention	behavioural	Give feedback to the fake patient
Clarity	is not clear and information is	is not clear and information is	Communication is not always clear and information is not always precise	precise but communication is	Information is correctly

Table 4.4 The OSCE assessment grid.

The acceptable level of performance is +2 points and corresponds to 18/30. The maximum score, instead, is +8 points and corresponds to 30/30. The OSCE examination is made by almost two tutors. To ensure the validity and reliability of the OSCE examination, it is important that all of the elements are well-prepared and carefully developed (6).

# 5.5 Emotional Competencies OSCE examination scenario for the role-play

Topic: the role of emotions in the caring relationship with the patient

#### **Overview**

1. Description of the situation to be reproduced

Mrs. Maria, aged seventy and suffering from diabetes, was admitted to the diabetology department for diabetic decompensation from oral hypoglycaemia. Mrs Maria was accompanied by her son to the emergency room two days ago due to tiredness, polyuria, polydipsia and general malaise. They have just informed her that she will have to change her therapy: she will take insulin therapy 3 times a day, after blood glucose measurement. You go to the patient who has just called for assistance. Mrs Maria is very worried about her situation and wants to go home as soon as possible.

- 2. Target: Nursing student II year.
- 3. Definition of participants' roles:

Elderly patient (co-protagonist): a nursing tutor

- Nurse (protagonist): Nursing student
- 4. Responsibilities of the actors: roles and activities

Patient: worried and uncertain about the new therapy.

Nurse: Establishes a relationship of trust with the assisted person, trying to reassure him/her about the management of the therapy at home.
#### <u>Aim</u>

- 1. Main objective: To ensure evidence-based care intervention that meets the needs of the person being cared for in the clinical setting.
- 2. Intermediate objectives:
  - To develop a relationship of trust with the person assisted:
    - Maintaining an open and respectful conversation with the person assisted
    - Taking the necessary time to establish a meaningful therapeutic relationship
    - Showing understanding and availability
    - Assessing how the context and environment may affect the development and maintenance of trusting relationships
    - Paying attention to aspects of non-verbal communication
  - To manage the emotionality of the person assisted:
    - o Identifying the assisted person's fear that hides behind his/her aggressiveness
    - Identifying the cared-for person's concerns
    - Explaining the importance of the correct intake of therapy
    - Sharing relevant information with the person assisted
    - Using appropriate verbal communication, paying attention to the tone of voice, using terminology appropriate to the person's ability to understand
    - Evaluating feedback on the patient's knowledge
  - To manage one's own emotionality:
    - Identifying one's negative emotions to manage them (anger, irritation, ...)
    - o Controlling one's emotions with an active and empathic attitude
    - Maintaining proactive and educational listening with the person assisted

<u>Environmental realism</u>: The environment should reproduce a room in a hospital ward, with classic hospital furniture (bed, chair, bedside table, wardrobe, ...). The actors must impersonate the roles accurately, through the correct management of clothing (clothes, ...).

<u>Complexity</u>: The scenario presents an ordinary situation of taking care of the assisted person. The complications concern the emotional situation of the person assisted, which could affect the adherence to home care.

#### Instructions for the person who holds the position of the patient

You are 70-year-old Maria, who has suffered from diabetes for seven years. You went to the emergency room unwillingly after your son, concerned about your situation, insisted. A week before admission, you felt very tired, so much so that you often sat watching TV and couldn't finish watching your programmes, you had to urinate often, but you think it was because you were drinking a lot. In the ward, the diabetologist said to stay calm, that your discomfort was due to a glycaemic decompensation but that with insulin therapy you would soon be back home, but you don't know them and they don't know you. They changed your therapy, but you are afraid of the injections, you don't know why you had 'glycaemic decompensation', and the doctors don't understand your situation.

<u>Spontaneous information</u> (to give to the students): When you receive the information about the change of therapy from oral hypoglycaemic to insulin, you are very afraid. You have always been afraid of injections and needles, and you don't know how to do them. You call the nurse to ask for clarification, but you are very nervous.

<u>Non-spontaneous information</u>: I am afraid of needles; I am afraid of not being adherent to the therapy because I have always been comfortable with the pill. The doctor was hasty in giving me the news, and for this, I am very angry because the hospital is a place of care and not a place to give hasty information.

#### Psychological experience:

- State of agitation and worry about the change of therapy
- Worried about the worsening of the illness
- Agitated about the doctor's lack of attention

#### References

- 1. Treccani Enciclopedie on line, Istituto dell'Enciclopedia Italiana. <u>https://www.treccani.it/vocabolario/valutazione/</u> last access: 22 September 2023
- Sasso L., Lotti A., Gamberoni L. Il tutor per le professioni sanitarie. Carocci Faber, Roma, 2003.
- Solà-Pola, M., Morin-Fraile, V., Fabrellas-Padrés, N., Raurell-Torreda, M., Guanter-Peris, L., Guix-Comellas, E., & Pulpón-Segura, A. M. (2020). The usefulness and acceptance of the OSCE in nursing schools. Nurse education in practice, 43, 102736. Advance online publication. <u>https://doi.org/10.1016/j.nepr.2020.102736</u>
- 4. Wosinskic J, Belcherb AE, Dürrenbergera Y, Allina AC, Stormacqa C, Gerson L. Facilitating problem-based learning among undergraduate nursing students: A qualitative systematic review. Nurse Educ Today. 2018;60:67–4. <u>http://dx.doi.org/10.1016/j.nedt.2017.08.015</u>.
- Nastos, S., & Rangachari, P. K. (2013). The TRIPSE: a process-oriented exam for large undergraduate classes. Biochemistry and molecular biology education : a bimonthly publication of the International Union of Biochemistry and Molecular Biology, 41(3), 145– 155. <u>https://doi.org/10.1002/bmb.20696</u>
- Bagnasco, A., Tolotti, A., Pagnucci, N., Torre, G., Timmins, F., Aleo, G., & Sasso, L. (2016). How to maintain equity and objectivity in assessing the communication skills in a large group of student nurses during a long examination session, using the Objective Structured Clinical Examination (OSCE). Nurse education today, 38, 54–60. <u>https://doi.org/10.1016/j.nedt.2015.11.034</u>
- Rushforth H. E. (2007). Objective structured clinical examination (OSCE): review of literature and implications for nursing education. Nurse education today, 27(5), 481–490. <u>https://doi.org/10.1016/j.nedt.2006.08.009</u>
- 8. J.-J. Guilbert (1990). Guide Pédagogique Pour les Personnels de Santé. World Health Organization, Geneva, Switzerland.
- McWilliam, P., & Botwinski, C. (2010). Developing a successful nursing Objective Structured Clinical Examination. The Journal of nursing education, 49(1), 36–41. <u>https://doi.org/10.3928/01484834-20090915-01</u>

## CHAPTER 6 The Emotional Competence VEIGA Scale (ECVS)

Authors:

Maria Augusta BRANCO, Lisa GOMES, João CAINÉ, Paulo NOVAIS, Rui PEREIRA, Filomena SOARES,

#### 6.1 Introduction

#### **Emotional Competence... a Conception in progress**

In my commitment to Life, I feel close to the need for reflection... with my eyes on myself, in communion with Spinoza's God ...

#### Perspectives for an Introduction

In general, workspaces, whether in scientific, technological, and mathematical fields like engineering, neuroscience, or aeronautics, or in humanistic areas such as social work, law, education, or healthcare, increasingly require individuals to possess soft skills that express their emotional competence. The behaviours needed to be assertive have become more complex, interrelated, and demand a combination of multitasking and professionalism, while still requiring a touch of charm and captivation, whether in one's personal or professional life.

The need to manage behaviours that express emotional body states is legitimized, essential, and in an underlying way. This territory concerns the dimensions of emotional competence, namely self-awareness and emotional management. Employers require a CV of empathy, self-motivation, drive and focus on the task. This current concept of professionalism is being expressed in all areas of work in general, and in health in particular.

The 21st century assumed organizations, in general, and health organizations in particular, as spaces for co-creation, drawn in flows of transdisciplinary interactions, designed in a paradigm that seeks quality and improvement. The current paradigms of excellence seek to find executory and evaluative standards, such as behaviour charts and work routes, in the sense and meaning of achieving that excellence, which is why their actors must have a sense of subjective well-being and promote their health through Emotional Education.

In the social and healthcare fields, especially in contexts involving pandemics, wars, or social confrontations, the fragility of human beings necessitates more comprehensive and diverse care approaches. In these situations, the concept of humanity becomes essential as it helps recognize the unendurable aspects of suffering, expressions of anguish, exclusion, and revolt. Caregivers in such contexts must possess emotional dexterity to effectively navigate these sudden, exhausting, and overwhelming situations.

So, in workplace contexts caregivers work in a setting of intellectual and work positions of and for excellence. In this broad context, it is worth placing the central idea that, in the binomial organizations and quality - in terms of human provision - the health-promoting training component should be underlying the holistic perspective, to which Emotional Education, and according to Pérez, et al. (2020) (1) does not omit itself. Thus, quality organizations and Emotional Education can be symbiotic. That is, it is assumed that this binomial will be expanded and promoted if this triangulation is present.

In this perspective, the institutions in general, and the health institutions, in particular, and within these, the organizational managers of the organic units and respective health services, who rescued the quality of care as the priority element of this paradigm co-creator of excellence, in the route of the visibilities, success and recognition of these institutions, are now called to objectify this triangulation, where not only, but also, according to the study by Almeida et al., (2021) (2) the concept of Emotional Competence is inserted.

From the point of view of operationalization, more than a starting theoretical and static construct, it starts with the planning of a transversal strategic axis, from the government entity to the operators in health institutions, capable of identifying units of care behaviour, building monitoring and evaluative attribution plans, to improve the quality of the services provided.

In this process, all resources are relevant and appropriate, especially human resources in an absolutely essential way, because they are the core elements of health teams, which also include nurses. And in them, their characteristics of productivity, knowledge and ability to comply, creativity and, above all, intra and interpersonal interaction.

This context is the exchange of energies, interactions, service provision, which serve and express the acts of care, at a bio, psycho, social, cultural and spiritual level.

The essence of the theme is nursing in action. Within this action, lies the implicit ability to care for oneself, to tend to one's internal world, and nurture one's inner milieu, all in the pursuit of holistic care for the external world. This self-care is perhaps the initial indicator of quality—taking care of oneself.

And in this context, it is important to learn/develop Emotional Competence, in its three self dimensions, at the intrapersonal level: Self-awareness, Self-motivation, Emotion Management; and at the interpersonal level: Empathy and the Management of Emotions in Groups, since it has been proven that "existing equivalences when relating the general competences of the work of nurses with the emotional competences and abilities that emerged from the responses of these professionals..." as we can see in the study by Carício et al. (2021), (3).

The body and/or cognitive expression, in conscience, of feeling emotionally competent, in fact translates an inner harmony, and, therefore, a level of availability to the outside, to the other, which can prove to be a sensitive indicator of the quality of care, and therefore quality in health, which, as defended by Donabedian (2003) (4), is strengthened in the interpersonal relationship, between client and health professional, by incorporating the principles of respect for the client's values, beliefs and desires and where the client is the centre of attention.

In fact, already at the beginning of the century, Donabedian (2003) (4) in the initial approaches to the formulation of the concept of quality in health, placed it as a structure supported by two sub-structures: (1) the product of the interaction of science and technology of care factors, and (2) the surrounding conditions.

And in this structure, he considered: a) technical-scientific dimension, b) the interpersonal dimension, the interpersonal relationship, health professional and client; c) and the environmental dimension. The author argued that quality in health would result from the way these variables are related, being considered, therefore, by the author as the central elements of the conceptual definition of quality in health. And, in that quality, the emotional interaction, and the perception about that, is a very pertinent issue to reflect about.

In these contexts, the relational and emotional field care, make sense to think about soft skills in general and about Emotional Competence in particular. In fact, when we say Emotional Competence, we describe the ability a person or a team express - or has to express - his or her own emotions with complete freedom, in what part or parts of the body, in what intensity, balance or during of time. The concept, is derived from the concept of emotional intelligence, which seems also to be the ability to identify emotions, and that is present in all human and animal life.

More specifically, when we talk about Competence, we want to talk about the level of skill with which someone interacts constructively with him or herself or other people.

Thus, and because this concept deals with understanding and/or recognizing at what level someone can achieve effectiveness and/or efficiency in a certain dimension, - such as Empathy or Emotional Regulation - it will be presented how the transformation of the concept was developed in construct and the respective operationalization of each of the five dimensions, as well as and necessarily the final operationalization of the global CE.

#### 6.2 Emotional Competence

#### Before the Concept, an overview of Emotion

The emotions of human beings are activated in the brain in response to different stimuli, which can be generated internally and externally, triggering a complex state of the organism, which is characterized by disturbance or excitement, which anticipates an organized response.

As we can see, in study by Stanojlović, et al. (2021), actually, research shows that the four components that influence each other represent the concept of emotions and complete the overall emotional experience, and these are: autonomous (the autonomic nervous system signals, increase in heart rate, blood pressure); somatic (body language, facial expressions); cognitive (control, management), and subjective feeling (emotion, individual experience). The interplay between emotions and cognition has been the subject of research. In this study, the authors, present the actual concept that *"Emotions can be evoked reflexively by simple physical stimuli (bottom-up), but can also be complex reactions involving cognitive, physiological and behavioural reactions (top-down). The amygdala, the <i>"alert" or "neural alarm" structure, is responsible for conditioning fear, while the medial prefrontal cortex participates in emotion self-regulation and decision making"*. (Stanojlovic, et al., 2021, p.361) (5).

This current proposal corroborates in terms of knowledge and conceptualization, what was previously described in research into Damásio (2000) that always likes to conclude with the idea that "*consciousness and emotion are not separable*" (6), but he always starts from the assumption that emotion is assumed to be a phylogenetically constructed program, and "...its biological impact is considered very different in terms of three phenomena that, despite being distinct, are closely related: *having an emotion, feeling that emotion, and becoming aware that we are feeling that emotion...*" (6) and how these different levels can be present in the person.

Indeed, we can have an emotion and not feel that emotion, and therefore not be aware that we are feeling that emotion. This fact can change the level of self-awareness of a person, who cannot, for that fact, be responsible, much less "guilty". This is a property that is developed with and during training in Emotional Education, and which, in its proven capacity, brings people closer to their Emotional Competence. Such skills are particularly important in nurses.

It is important to clarify, since the issue of providing care and services is addressed here in a context of greater or lesser expression of suffering, pain and/or anguish, that any of these phenomena may or may not trigger emotions in nurses.

As the author explains: "... human emotion (6) is triggered even by music and by cheap films, whose power we must never underestimate. The human impact of all these causes of emotions, (...) depends on the feelings engendered by these emotions. It is through these, which are private and directed inwards, that the emotions, which are public and directed outwards, begin their impact on the mind; but the full and lasting impact of feelings requires awareness, for only in conjunction with the advent of a sense of self do feelings become known to the individual who has them" (6).

And Damásio continues (6) "... I separate three stages of processing that are part of a continuum: a state of emotion, which can be triggered and executed unconsciously; a state of feeling, which can be represented unconsciously, and a state of feeling made conscious, i.e., which is known by the organism that is having emotion and feeling."

It is important to make clear the phenomenon of "... contrast between the condition of emotion, which is induced without our knowing and turns outwards, and the condition of human feeling, which is essentially known and turns inwards..." (6), can leave us completely confused and unprepared to face this

phenomenon in us, that others know about me through my expression, than I through the feeling I feel about that emotional expression to which others have already acceded.

This point is essential to understand the value of learning self-awareness – the first dimension of Emotional Competence - from body self-awareness... because, as Damásio explains, "... like emotion, conscience is based on the representation of the body..." (6), thus assuming that the body assumes the status of an entity, capable of elaborating its own speech and emotional expressive grammar, and that somehow, what the body says/expresses says nothing better.

#### 6.3 The Emotional Competence Dimensions - Operational terms

#### Self-awareness

It is the first dimension and emerges from a corporeal self-awareness, produced from a fluid dialogue between the body and the somatosensory systems, where mental maps are constructed and reconstructed in a total and progressive immediacy. In fact, the essence of our consciousness is the body mapped in the brain, in the totality of the bodily structures in functioning or not, in a total and absolute precision. This process, based on the expression of Damásio (6) "...by mapping your body in an integrated way, the brain manages to create the crucial component of what will become the self»,(6), and therefore a current and dynamic self, updated through networks of chemical and neural signals, where arms and legs, viscera and vascular networks are completely and in detail mapped – positions, movements, blood pressure, blood pH, artery tension, irrigation hollow viscera, hemodynamics, stasis points, endocrine and exocrine secretions, in short... - in their respective changes at every moment.

Everything and about everything, body and brain exchange dialogues, and "... the brain is able to do more than map states that are occurring (...), it can transform body states, simulate body states that have not yet occurred...» (6). In short, the body says to the brain: "this is how I am constituted and this is the state I find myself in now" (6) and the brain, which even receives qualitative orientations about the "states of the body", guides the body in the functions to be maintained in the sense of homeostasis, and whenever requested "... the brain tells the body how to create an emotional state" (6). Thus, from the point of view of operationalization, the concept of self-awareness concerns the scrutiny and recognition, in consciousness, of any changes in these body states, triggered from an emotional trigger, or taking this activation (arousal) because of an Emotionally Competent Stimulus (ECS) to the subject.

#### Emotional Management

This dimension, in Emotional Competence concept, has been the subject of progressive studies and therefore progressive changes in its definition and conceptualization. From the beginning of this century with Damásio (2000) (6), and some new perspectives for understanding how the body integrates and manages emotional body states (2010) (11), until the recent research results, namely the presented by Myruski et al. (2022) (7), which unequivocally explain that the key feature of adaptive Emotion Regulation

is the capacity to manage behaviour during a range of emotional challenges. Other literature (Gross, 2015) (8) and other researches, also present for example, the managing frustration and delaying gratification, as common regulatory challenges in childhood – but we can have it in adulthood - , and, people in general and children in particular, all of us along life, can vary in their and ours use of adaptive or maladaptive Emotions Regulation behaviours during these challenges. So, recent authors when assume Emotional Regulation, assume the behaviour regulations in the arousal experience of an emotion.

Based on this assumption and current publications, this dimension starts from the recognition that the act of "management" does not refer to emotion, as an *arousal*, as the autonomic *activation*, but rather and only to the behaviour resulting from this activation. It is important to clarify that the basic emotions, Jamesian, of a sudden and/or unconscious event, are not considered, a priori, an object that can be managed. Even in its late arousal, after recognizing and mapping this body state, at a somatosensory level, it is not considered exactly manageable at a behavioural level spontaneously.

When we're talking about Emotional Management, what is assumed here is the management - scrutiny and recognition of signs and body modifications and the consequent interaction of the subject, consciously, about this process -, of the behavioural component of secondary and/or social emotions (7), and/or any kind of emotional feelings expressed about emotional body states emerging from an ECS.

The interaction of the subject on his own observation of the emotional occurrence (or emotional feeling), in the sense of his objectives, without suppression, nor loss of homeostasis, of his inner milieu, nor external context, is considered here, management of emotions. It is important to clarify that the so-called regulation of emotions (self-regulation) and cognitive flexibility (Decety & Jackson, 2004)(9), emerge from self-awareness and awareness of the other that develop simultaneously in preschool years and play an important role in this dimension. Its pertinent to present the Langeslag and Surti (2017) (10) perception, defending that we can control our emotions to some extent. But, they also argue that emotion regulation is the use of behavioural or cognitive strategies to generate new emotions or to increase or decrease the intensity of current emotions (10).

In scientific publications we can find some approaches to multiple regulation strategies, including for example, the distraction and or the cognitive reappraisal. Distraction involves thinking about something unrelated and reappraisal entails changing the meaning of a situation by reinterpreting it (10). Reinterpreting the situation can, for example, be done by changing the personal relevance, which has been called self-focused reappraisal, or by imagining better or worse outcomes, which has been called situation-focused reappraisal (9).

Langeslag and Surti (10) advice to reflect about the fact that it has been proposed that the effectiveness of reappraisal to up- and down-regulate emotions may vary depending on the intensity of the emotion, but this has not been systematically tested yet, suggesting that research must continue and results must be reflected, and Ochsner & Gross, (2008) (11) that present the concept and the skill of cognitive emotion regulation, as a kind insights from social cognitive and affective neuroscience, that we can learn and understand.

#### Self-motivation

In this operational context, the concept of motivation is integrated in the sense of "genomic unconsciousness» (12), and in the words of Damásio "...a series of behaviours that may appear to be decided by conscious cognition, but that are driven by non-conscious dispositions." (12), and, in order to situate this construct at the level of brain structures, explains that they will be "unconscious bases of behaviour, within the scope of instinct, automatic behaviours, impulses and motivations" (12). In terms of definition, it would be assumed that the concept of self-motivation relates to recognizing the basic elements of the energy that triggers or "moves" a certain attitude or the pursuit of a given behaviour: whether autotelic, or exotelic, and how these two limit points are recognized in the person. But not only that, we also intend to assess what limits and differences the person perceives, between (1) the assumption of a frontal and permanent attitude, (2) the obstinate repetition, (3) the change of the route without feeling of loss or damage, (4) procrastination or laxity.

#### Empathy

It starts with the concept and function of "mirror neurons" (6), (12) to access the skill of producing "as if" phenomena in the body, to understand and defend the cognitive component of empathy, which is characterized by the capacity to accurately infer the feelings and thoughts of another person, without necessarily experiencing the same feelings as that person. This point is here of the greatest importance. Such a perspective involves neutrality and impartiality of the one who empathizes and has been called perspective taking (Davis, 1983 (13); Ickes, 1997(14), a phenomenon that had already been identified, if there is a dynamic interrelationship (Decety and Jackson, 2004 (15) between the "I" (self-awareness) and the "other" (awareness of the other). The ability to represent and report one's own mental states is what will allow inferential knowledge of the mental state of others.

It is assumed that empathy in human beings is an innate attitude with four dimensions, and in the recent publication (Hoffman, 2000) (16), corroborated that the cognitive component, allows us to understand what Damásio (2010) (12) presented "...the actions of others, by putting ourselves in a comparable bodily state, (...) when we see the action of another person, our somatosensory brain adopts the bodily state that we would assume if we were moving..." (12), clarifying with some specificity that it is "...the connection we establish between our body states and the meaning they have assumed for us, which can be transferred to the simulated body states of others (...) we can assign a comparable meaning to the simulation." (12).

The affective dimension concerns the cognitive understanding of the feelings and emotions of others, and later allows one to empathize with empathy, in the depths of one's being, the emotions associated with something, whereby it is possible to internalize and personally experience the suffering suffered by others. The motivational dimension is the phenomenon of generating altruistic and interpersonal empathic motivation - feelings of compassion for others - that we observe or recognize. The prosocial dimension concerns how this compassion encourages us to be proactive, taking prosocial or ethical actions in defence of or for something or someone.

In the recent research, Decety & Jackson (2004) (15) presented studies shown that only part of the network mediating pain experiences (including the anterior cingulate cortex and the insula) is shared when empathizing or evaluating the pain in others. Most importantly, the authors defended that empathy - as we know the concept - also necessitates emotion regulation for which the ventral prefrontal cortex, with its strong connections with the limbic system, dorsolateral, and medial prefrontal areas, plays an important role. Once again, we do not assume that there is a unitary empathy system (or module) in the brain. Rather, we consider multiple dissociable systems to be involved in the experience of empathy. Finally, as suggested earlier, empathy is a motivated process that more often than commonly believed is triggered voluntarily. This makes empathy a flexible human capacity as well as a method of gaining knowledge of understanding another, and it is susceptible to social-cognitive intervention, such as through training or enhancement programs for targeting various goals"(15). As we can see, the concept is more complex and deeply inside interaction between body and brain structures.

#### Emotional Management In Groups

It is the dimension of the E.C. unifying empathy and emotional management, and which concerns the already mentioned "empathic" perspective-taking, that is, the phenomenon already identified by Decety and Jackson, (2004) (15), if any, and when there is a dynamic interrelationship between the "I" (self-awareness) and the "other" (awareness of the other). This dimension examines the effects of empathy, emotional contagion and emotional management.

It emerges from plural empathy, in which the absorption of the emotional expressions of others are perceived, and the respective impacts on us trigger emotions and emotional feelings to which we manage to give a response and behavioural expression, within an intellectual organization, which keeps us under control, and which, in turn, we may be able to spread to others. Previous studies about the process model of emotion regulation Ickes (1997) (14), has previously demonstrated that antecedent focused emotion regulation results in improved outcomes compared to response-focused emotion regulation when people are attempting to regulate their own feelings.

Reeck and Onuklu (17) in their study about Interpersonal Emotion Regulation (IER), the Consequences for Brands in Customer Service Interactions, they start from studies also demonstrate similar benefits of antecedent-focused emotion regulation when it is applied interpersonally.

These authors focus not only on the consequences of the strategies people use to manage their own emotions, to instead examine the consequences of the strategies people use to manage others' emotions. So, the current research builds on prior demonstrations that IER can improve relationships between people by showing that these benefits can not only emerge between these social and personal interactors as with some resources or contexts. Reeck and Onuklu (17) corroborant previous results (16), demonstrating that successfully managing negative others emotions can result in improved trust and best interactions. More than these, and taking prior demonstrations (Niven et al., 2012 (18), Bloch et al., 2014 (19) that IER between people results in increased trust and relationship satisfaction.

#### **Operationalize Emotional Competence**

Emotional competence has continued to be presented, in the literature, in a similar way to the concept of Emotional Intelligence, but each one of them has a different perspective on capabilities related to emotion, different methodologies of development and application, and different instruments for data collection and analysis of results.

The literature reveals that sometimes they are carelessly applied as synonyms (19), causing theoretical and methodological confusions, (19). Recently, Pérez-González et al (1) in an analysis of previous works, report Roberton et al. (2014) (20) definition of emotional competence as "...the ability to deal effectively with emotional information, that is, with one's own feelings and desires", and argue, quoting Davis & Qualter, (2020) (21), that emotional competence:

"...has been used as an aggregating term and as a multifaceted construct that encompasses emotionrelated skills (e.g., emotion awareness, emotion perception, emotion regulation), dispositional skills and qualities (e.g., emotional self-efficacy), which enable the individual to function effectively in emotionally charged situations." Pérez-González, (1)

In a very brief summary, Emotional Competence (E.C.), as an autonomous concept, seems to have emerged with the mixed models of Emotional Intelligence (E.I.), such as those of Goleman and Bar-On, but in fact it did not happen that way. Two authors in different countries – Saarni (22, 23) in the USA and Bisquerra (24, 25) in Spain – developed the construct of Emotional Competences (in the plural), more focused on the contexts of education and youth training, in order to make emotional education training in higher education more understandable and adaptive, as some authors advocated (Fragoso-Luzuriaga, 2015) (26) to develop students' emotional competence, in those contexts.

Later, in Portugal, studies by Veiga-Branco (2004) (27) e (2005) (28) on teachers appear, which presents the concept of E.C. based on the mixed model of E.I. (Goleman, 1995) (29).

Veiga-Branco (27, 28, 30, 31) studies Emotional Competence in the context of teachers and needs of the teaching-learning process, and starts the work, based on the modified concept of E.I. (32), modification that operated when perceiving the added value of E.C., reflecting what the author explores in *Notas* (32): "... the expression "emotional competence" includes both social and emotional competences, clarifying that "an emotional competence is a learned ability, (...) that results in an extraordinary performance at work" (32), where he pointed out that:

"Our E.I. determines our potential to learn practical skills that are based on five elements: Self-awareness, motivation, self-mastery, empathy and talent in relationships. Our E.C. shows the extent to which we translate that potential into professional skills".

Thus, later (28) distinguishes this concept from E.I., arguing that "the E.C. exists when someone reaches a desired level of achievement", and that this concept concerns the post-facto. It can only be appreciated simultaneously or after the exhibition of behaviours and/or attitudes, through observation or through the expressed memories, by executing subjects or by observers.

Starting from these elements as a theoretical framework, and having the exploration, identification or recognition of a E.C. profile as the object of study, it creates an Analysis Instrument: the Veiga Branco Scale of E.I. (EVBCIE) (27, 28, 30, 31).

This instrument was subsequently reformulated, to an instrument with 86 items - the Emotional Competence Veiga Scale, (EVCE), used in progressive studies, in successive samples, since 2010, until two years ago, for exemple, the "Emotional Competence profile in a higher education students" (Veiga-Branco, 2011, (33), the study in "Emotional competence in healthcare professionals: correlational and comparative research" (Veiga-Branco et al, 2012) (34), and the study about "Emotional competence (EC) in physiotherapists in the national network of integrated continued care" (Afonso, et al., 2013) (35), and even out of health scientific area, the study developed in Emotional Competence in Portuguese political decision-makers in Veiga-Branco & Costa (2015) (36).

This data collection instrument, with 86 items, has been progressively under development towards its reduction, resulting in the study of the reduced form, the Emotional Competence Veiga Scale (ECVSr33), in the original Portuguese version: *Escala Veiga de Competência Emocional (EVCEr33)*.

This current reduced version of the Scale, has already been used in some studies, with the respective application and statistical study in several samples, namely in the areas of Health, Education and Finance, as the nursing context in Almeida et al. (2021) (37) and even in Moreira, et al. (2021) (38), and in local political decision makers in the study by Nogueira et al. (2021) (39) having the respective statistical studies, considered the EVCEr33, a measuring instrument with the capacity to study the EC profile, in different areas.

Among others, the profiles of health professionals (34, 37, 38), and recently, the results by Rodrigues, (2017) (40) and by Ferreira (2018) (41), we have more recent results, in recent studies corroborating some EC dimensions values, as the study by Giehl et al. (2020) (42) and the last one by Silva et al. (2023) (43) in Management and Finance contexts. In terms of results, successive correlational studies, with analysis of principal components, have been showing the profile of E.C. through the five dimensions, - self-awareness, management of emotions, self-motivation, empathy and emotional management in groups -, which may be differently correlated with each other and with the E.C., depending on the contexts of the samples.

#### 6.4 Measuring Emotional Competence

To study a phenomenon, primarily non-measurable, the research methodology is clear: it is necessary to transform the concept into a measurable phenomenon, then the phenomenon into a variable and finally operationalize the variable at either a quantitative or qualitative level.

And, at this point, it is important to consider Emotional Competence dimensions, as skills in the phoneme quality (of life), that is recognized as an adjective and therefore subjective as a concept, but in this context, it is considered as a variable to be operationalized, to become the evaluable product of an instrument to be applied.

This instrumental perspective of quality (of life) has to reveal an appreciative and measuring capacity of actions, activities and planning, of decision makers, executors and evaluators, and of all those who interact and integrate the prescriptive and care setting of all health teams.

The necessary transformation of a concept into a construct, and a construct into an evidence-instrument, is quite curious: in fact, it is practically a matter of obtaining the measurement of the quality of actions, through what was identified as acting *quality indicators*. And it will be these elements, in their most elementary form, simplistic or not, that portray and express the signs related to the structure, processes and results of preconceived actions, at the time of formulating the objectives, and that operationalize the concrete evaluation that promotes the quality of "to know to be" in emotional competence interactions in life.

In any life context – personal, social, and work – each of the primary independent variables of Emotional Competence (EC), represented by its five dimensions, can assume the status of behavioural and attitudinal variables of the five core concepts within the construct of Emotional Competence. These concepts reflect behaviours at the relational and inter-relational levels, making them valuable tools for analysing caregiver-user relationships, aligning with attributes such as effectiveness, efficiency, optimization, legitimacy, acceptability, and equity. These attributes, when used and measured in isolation or in combination, contribute to the meaning of 'quality in health' (44), particularly in the context of professional caregivers, including nurses, doctors, lawyers, and military personnel.

It would be both interesting and relevant to leverage the behavioural and attitudinal elements that comprise Emotional Competence to explore certain components of professionalism. These behaviours and interactions are inherent in achieving desirable relational quality. For instance, correlating the dimensions of Self-Awareness, Emotion Management, Self-Motivation, Empathy, and Emotion Management in Groups (teams) with the required functions and skills in various fields of work.

To assess whether individuals exhibit these behaviours in practice, one can simply review the 33 items comprising the Emotional Competence Scale, which effectively represent each of the five dimensions. By doing so, individuals can determine whether these behaviours are present in their lives, whether they have occurred in the past or are ongoing. This scale provides a practical repertoire of attitudes and actions that individuals can access, irrespective of the context.

#### Every road has a story... what about EVCE road?

The instrument, or the Data Collection Instrument (DCI) that has been used so far, the "Emotional Competence Veiga Scale (ECVS), (EVCE) (28, 31, 34,45) includes 86 items to operationalize the five dimensions of E.C.. The first dimension, Self-awareness was composed by 20 items, the Emotion Management dimension was composed by 19 items, the third dimension, Self-motivation, integrated 21 items, the fourth dimension, Empathy was constituted by 12 items, and the last one, the fifth dimension Management of Social Emotions was constituted by 14 items. These items are constituted as ordinal variables, and each of these items was, in turn, operationalized through a scale measured by a Likert-

type temporal frequency scale (1 to 7, with 1 - "never", 2 - "rarely", 3 - "occasionally", 4 - "usually", 5 - "often", 6 - "very often" and 7 - "always", with participants being able to choose how often they experience the situations described.

The Psychometry of the Data Collection Instrument (DCI), and from the point of view of descriptive statistical analysis, assumes that the EVCE cut-off point is 4. To measure the EC profile, 3 levels of E.C. are considered: the **low level**, between 1 and 3.49; the **moderate level** between 3.50 and 5.45, and the **high level** between 5.46 and 7.

Therefore, in order to obtain the level of Emotional Competence as a global construct, we will have to obtain the average values of the means of the five dimensions. To obtain the level of each of the dimensions we have to obtain the average values in each one.

The actual scale, EVCEr33, as Data Collection Instrument (DCI) and that has already been used, (37, 38, 39, 42, 43) includes only 33 items to operationalize the five dimensions of E.C., as the following.

- The first dimension, Self-awareness is actually composed by 8 items, one of which, is an inverted item;

- the second one, the Emotion Management dimension is now composed by 7 items, three of which, are inverted items;

- the third dimension, Self-motivation, now integrates 7 items, two of which, are inverted items;

- the fourth dimension, Empathy, is actually compose by 5 itens, four of which are inverted items;

- the last one, the fifth dimension Management of Social Emotions actually integrates 6 without inverted items.

For example, and to better understanding, the level of Management of Social Emotions dimension we have to obtain the average values of those 6 items, and if the average is between 3.50 and 5.45, we can assume thar our sample present a **moderate level** in **Management of Social Emotions**.

As the previous DCI, in the actual, all items are also constituted as ordinal variables, and each one is also operationalized and measured by the same Likert-type temporal frequency scale, and is also applied the same Psychometrics.

Therefore, and as has been explained, the analysis of the Instrument has to consider the normal configuration items and the inverted items, that must be assumed in a reverse way, among the total items of the EVCE. From the point of view of constructing factor analysis: the items related to each dimension of the E.C. must be submitted to Principal Component Analysis with varimax rotation, whose set of extracted factors aggregates a certain profile in each dimension of the E.C.. The scale was successively and progressively applied and after each use, the results were subjected to exploratory validation of the instrument, through analysis in main components (SPSS), so, in a first time, for the

Portuguese population (studies published and carried out in Portugal: two with teachers (27, 28) and three with nurses: Ferreira (2018) (41) studied the Formal and Informal caregivers in Primary Health Care Units, Veiga-Branco & Lopes (2012) (44) studied the Nursing Teams in Continued Care National Network, and, some years before, Agostinho, 2008 (45), had previously studied hospital nurses, through Principal Component Analysis.

In the successive application of the Scale, it was verified that the successively and sequentially obtained internal consistency values ( $\alpha$  Cronbach), assume the validity of the DCI and provide a rigorous statistical treatment. find a level of quality for the successive Emotional Competence profiles of successive samples, and through exploratory factor analysis, we can have a summarized perception of this profile. The successive and progressive results obtained made it possible to access the respective construction of the averages analysed and the predictive variables of E.C. and that led to the construction of an E.C. profile.

In the Serious Game and Emotional Competence studies, we had an analyses, where we can observe the following in the text.

#### Data Analysis Methodology

The data analysis process took place in <u>two phases</u>. The first phase focused on the validity, reliability, and confirmation of the emotional competence scale used in the questionnaire. The second phase of the data analysis was intended to descriptively analyse the results obtained and identify statistically significant differences between groups.

#### <u>1<sup>st</sup> Phase</u>

It should first be mentioned that all the methodological requirements for carrying out the factor analysis of the emotional competence scale were verified. All variables are metrics (Likert scale), all factors have at least 5 variables and there are more than 5 observations per variable (Hair et *al*, 2010 (46)).

Confirmatory analysis was carried out using structural equation analysis (SEA). The use of SEA allows a qualitative leap from an exploratory model to a confirmatory model, going beyond classic statistics:

Another advantage of using this methodology is that the SEA allows testing the global fit of models and individual significance of parameters in a theoretical framework that includes various types of linear models.

The internal reliability of the scales was examined using Cronbach's alpha analysis. Cronbach's alpha is one of the most used measures for internal verification of a group of variables (items) and can be defined as the correlation expected to be obtained between the scale used and other hypothetical scales of the same universe, with an equal number of items, which measure the same characteristic.

#### 2<sup>nd</sup> Phase

In addition to carrying out the descriptive analysis of the variables using frequency distribution tables (in the case of the nominal variables), some measures were also examined, such as the mean, standard deviation, minimum, maximum, and median (in the case of variables of a quantitative nature).

To test the existence of statistically significant differences in the dimensions of emotional competence according to gender and the fact of being student workers, Student's t-tests were used for two independent samples. 1-way ANOVA tests were used to test the existence of statistically significant differences in the emotional competence dimensions according to the year of attendance and age groups. Since we are working with large samples, the assumption of population normality for carrying out these parametric tests was validated in most crosses by invoking the central limit theorem.

All statistical analyses were performed with IBM SPSS Statistics 24.0 and IBM AMOS (Chicago, IL)

#### Results

#### The list of EVCE-r33 items and the corresponding division by dimensions are presented in first place:

Table 1 – Items by dimension (Dim), Description in the Model (Discripti), Items, and Inverse Items

Dim	Description	Item Description	Inverse Item
Self-awareness	EVCE1A	I allow myself to be absorbed by these emotions, which I am unable to escape, and they end up conditioning my behaviour.	Yes
varen	EVCE1B	I have the exact notion of the type of feelings that invade me; e.g. if it's anger, fear, contempt hate, frustration I can define them.	No
ess	ECVE1C	Once invaded by negative feelings, I can't control them	Yes
	EVCE1D	Decreases my level of reasoning. I can't concentrate easily	Yes
	EVCE1E	My behaviour changes at the relational level I remain silent; euphoric, sulking, etc	Yes
	EVCE1F	I am mentally stuck in these feelings for a long time. That feeling always comes back	Yes
	EVCE1G	I'm the unstable type, with various mood swings	Yes
-	EVCE1H	I'm the Unlucky type, I'm unlucky in life	Yes
imo	ECVE2A	I did active physical exercise (aerobic). I spent the energy in activity.	No
Emotional Management	EVCE2B	I reasoned. I tried to understand and identify what led me to anger. I thought better of that situation	No
Mana	EVCE2C	I had to unload I tend to use objects, people or situations as targets for my anger, even though I feel less good about myself afterwards.	
ageme	EVCE2D	I live in a state of chronic worry about the event that caused my rage and I keep thinking about the words/people/attitudes that caused it.	Yes
ent	EVCE2E	Apparently peaceful Intrusive, persistent thoughts that haunt me day and night	Yes
	EVCE2F	I feel relief if I practice physical exercise or sport.	No
(0	EVCE2G	I feel relief if I eat or drink.	Yes
elf	EVCE3A	Pessimist (whatever I do, it will go wrong).	Yes
mo	EVCE3B	I'm able to get out of any trouble!	No
Self motivation	EVCE3C	Calm in life. I am completely absorbed in what I am doing, indifferent to what surrounds me	No
n	ECVE3D	Apparently calm, I'm doing what I have to do ruminating on the thoughts that occur to me	
	EVCE3E	Uninteresting Self-pity invades me. I just feel "down"	Yes
	EVCE3F	Resentful Contempt, resentment invades me. I cut ties with those who reject me, I resent that person(s)	Yes
-	EVCE3G	Who thinks: It went wrong as a result of a personal defect, that's who I am.	Yes
mp	EVCE4A	"Recording"/noticing the feelings of others.	No
Empathy	EVCE4B	"Reading" non-verbal channels (voice tone, hand gestures, facial expression, gaze direction, behavioural attitude, position, etc.)	No
	EVCE4C EVCE4D	Realize the consonance between the words and the body attitude of the person. Use calm (but consciously) to listen I realize that I feel good listening to people	No No
	EVCE4E	I become receptive to the instability of the other and trigger an unstable attitude. I get irritated.	Yes
Relat Vlan;	EVCE5A	I can understand how people are feeling	No
Relationship Management	EVCE5B	In my relationships with others, I clearly say what I think, regardless of the opinion they express.	No
ip ent	EVCE5C	I have control over my own feelings.	No
	EVCE5D	I can pick up on their feelings and I seem to start absorbing them.	No
	EVCE5E	I have an innate sensitivity to recognize what others are feeling.	No
	EVCE5F	I feel the interaction I feel physically synchronized with those around me.	No

Given this information, it was obtained the following model of structural equations with a view to confirmatory factor analysis as we can observe in the Figure 1.





After specifying and obtaining the confirmatory measure model, let's now examine some of its empirical indices that assess the goodness of fit of the obtained model:

After the statistical analysis considered for this purpose, it was found that: Virtually all indices refer to a model with a good fit. As we can observe in Table 2, only the RMSEA (*Root Mean square Error of Approximation*) with the value 0,083, refers to a Poor fit: ]0.5; 0.10[, as it tends to penalize this model due to its complexity.

Given these results, it is considered that there is no need to question the model, and it should not be rejected, and which is considered the acceptance with no need for refinements through an exploratory strategy to find a model that best explains the observed data. Thus, the conditions for using the EVCE-r33 scale in the sample are met.

Absolute indices	Value	Diagnosis
X2/df	2,718	Good fit: ≤ 2-3
Root mean squared residual	0,253	Closer to 0 than 1. Well rated model
Adjusted GFI	0,690	Good fit: ]0.6; 0.8[
Root Mean square Error of Approximation	0,083	Poor fit: ]0.5; 0.10[

 Table 2 – Distribution of the Absolute Indices of the Model quality assessment

As we can see in Table 3, the main characterizing traits of the sample of 241 nursing students, presents the sociodemographic variables, where we can observe that students present age with mean age of 20,48 years old, (standard deviation of 4.18 years), as maximum with 25 years old and the minimum with 17 years old, with the vast majority of respondents (82, 6%) being up to 21 years old. We can verify that the majority (n=215; 89,2%) are female students, and 151 (62,5%) attending in the first and second academic years, 73 students (30,3%) in the first and 78 students (32,4%) in the second academic year. When asked if they were student workers, only 30 (12.4%) of respondents answered in the affirmative, so the minority of the students who made up this sample were working students.

	n	%
Sex		
Female	215	89,2
Male	24	10,0
Other	2	0,8
Age categories		
Up to 21 years	199	82,6
More than 21 years	42	17,4
Year attended		
1 <sup>st</sup>	73	30,3
2 <sup>nd</sup>	78	32,4
3 <sup>rd</sup>	61	25,3
4 <sup>th</sup>	29	12,0
Student worker		
No	211	87,6
Yes	30	12,4
Age: X=20,48 years (SD 4	,18); Min: 17; Max: 25	

Table 3 – Distribution of absolute and relative values of the sample sociodemographic variables

#### **σ** (sigma) – Standard Deviation

The descriptive analysis of the 5 dimensions in study, presents, in Table 4, that the reliability results were satisfactory in almost all dimensions. Only the emotional management dimension has a Cronbach's alpha value below the acceptable value (0.498<0.600).

On average, the emotional competence dimension with the highest expression in the sample is "Empathy" (mean 5.02 and standard deviation of 0.81). On the opposite side is the "self-motivation" dimension with a mean value of 4.27 (standard deviation of 0.71).

But and considering the The Psychometry of the Data Collection Instrument (DCI), to measure the EC profile, and in the 3 levels of E.C. considered, we can observe the low level, between 1 and 3.49; the moderate level between 3.50 and 5.45, and the high level between 5.46 and 7.

In analysing Table 4, it can be seen that the students in the sample present a moderate level of effectiveness, whether in each of the dimensions or in the overall CE, since all the respective average values are between 3.50 and 5.45.

However, it appears that there is a part of the sample, which presents a low level of effectiveness, in any of the dimensions with minimum values between 2.13 (sd=0,91) for self-awareness, 2.14 for emotional management (sd=0,75) and self-motivation (sd=0,71), and for Relationship Management (X=2, 33; sd=0,79). Conversely, it also appears that some of the students presenting a high level of effectiveness in terms of the five dimensions, as we found maximum values the high level of (X=7.00) in terms of self-awareness (sd=0,91), empathy (sd=0,71), and relational emotional management (sd=0,79).

The highest level found was Empathy, with an average of 5.02 indicating that students present a moderate level in this dimension, and with a Cronbach value ( $\alpha = 0,743$ ) indicating that there is a good level of internal consistency between the items.

The lowest mean value (X=4,33) in the sample indicates that it has moderate level in Relationship Management but also the value of Minimum (2,33) also indicate that a part of students are in low level in that dimension.

1	No Items	α	Min	Max	х	σ
Self-awareness	8	0,851	2,13	7,00	4,41	0,91
Emotional management	7	0,498	2,14	6,57	4,51	0,75
Self-motivation	7	0,602	2,14	6,14	4,27	0,71
Empathy	5	0,743	3,00	7,00	5,02	0,81
Relationship Management	6	0,729	2,33	7,00	4,33	0,79
Global Emotional Competence	e 33	0,865	3,27	6,36	4,48	0,57

Table 4 – Values of Descriptive analysis of the EVCE-r33 dimensions and Global Emotional Competence

 $\pmb{\sigma}$  (sigma) sd -Standard Deviation;  $\alpha$  -  $\alpha$  de Cronbach

In global of the sample, we can verify a Moderate General Emotional Competence Female Profile (X=4,47; sd=0,55) and Male Profile (=4,65; sd=0,66), and with no statistically significant differences between (p=0,148) genders sample. Male profile is higher due to the significantly higher Emotional management profile.

There are statistically significant differences in the self-awareness dimension according to gender, with male respondents presenting a significantly higher mean (X=4,84; sd=1,06) and (p<0.05), different of female sample with average 4,38 (sd=0,88). Close to statistical significance (p=0.057) is also the highest mean value of those respondents in the Emotional management dimension, and female students with 4,49 (sd=0,75).

On the contrary, but without statistically significant differences, female sample present higher mean in the Empathy dimension (X=5,05; sd=0,81) and p<0.145, and Relationship Management dimension with mean 4,36 (sd=0,78), with respectively lower values of male sample with average 4,80 (sd=0,82) in the Empathy dimension, and Relationship Management with the mean of 4,24 (sd=0,84), but the both in the moderate profile.

	Female	e	Male		
		Standard		Standard	
	Mean	deviation	Mean	deviation	р
Self-awareness	4,38	0,88	4,84	1,06	<0,05
Emotional management	4,49	0,75	4,80	0,76	0,057
Self-motivation	4,25	0,69	4,52	0,77	0,073
Empathy	5,05	0,81	4,80	0,82	0,145
Relationship Management	4,36	0,78	4,24	0,84	0,484
General Emotional					
Competence	4,47	0,55	4,65	0,66	0,148

**Table 5** – Values of Mean and stand deviation (St.Dv.) of the five EVCE-r33 dimensions according to gender

 $\sigma$  (sigma) sd -Standard Deviation

Observing Table 6, we can see that both groups - students up to 21 years and more 21 years – present the higher profile in Empathy dimension, with youngers mean of 5,06 (sd=0,80) and the others with mean of 4,82 (sd=0,88), but we can see that in this difference, respondents aged up to 21 years have a significantly higher mean in the dimension related to empathy.

The lower profile of two groups students - up to and more than 21 years - is in Relationship Management with mean 4,37 (sd=0,79) and 4,17 (sd=0,76) respectively, without statistical significance (p=0,130)

	Up to 2	21 years	More th		
		Standard		Standard	
	Mean	deviation	Mean	deviation	р
Self-awareness	4,39	0,91	4,53	0,91	0,351
Emotional management	4,53	0,76	4,43	0,75	0,409
Self-motivation	4,25	0,72	4,38	0,64	0,284
Empathy	5,06	0,80	4,82	0,88	<0,05
Relationship Management	4,37	0,79	4,17	0,76	0,130
General Emotional					
Competence	4,49	0,56	4,45	0,61	0,734

**Table 6** – Values of Mean and stand deviation (St.Dv.) of the five EVCE-r33 dimensions according to age categories

 $\sigma$  (sigma) sd -Standard Deviation

It is observed, in Table 7 that the highest profile is in Empathy dimension for both groups, with mean of 5,04 (sd=0,80) to No student worker group, and to Student worker group the value of 4,90 (sd=0,94). Differently, the lowest dimension's profile, is in Self-motivation dimension (X=4,26; sd=0,72) for No student worker group, and for the Student workers, the lowest dimension's profile, is in Self-awareness dimension (X=4,35; sd=0,95).

But we can see that there are no statistically significant differences in the EVCE-r33 dimensions depending on being a student worker or not.

All dimensions, as well as the Global Emotional Competence, present average values that express a moderate profile, for both groups: No student workers with the mean of 4,48 (sd=0,56) and Student workers with the mean of 4,46 (sd=0,61).

**Table 7** – Values of Mean and stand deviation (St.Dv.) of the five EVCE-r33 dimensions depending on student worker

	No stude	nt worker	Studen	t worker	
	Mean	σ	Mean	σ	р
Self-awareness	4,42	0,91	4,35	0,95	0,680
Emotional management	4,53	0,74	4,40	0,86	0,363
Self-motivation	4,26	0,72	4,37	0,63	0,408
Empathy	5,04	0,80	4,90	0,94	0,457
Relationship Management	4,32	0,76	4,42	0,99	0,517
Global Emotiona	al				
Competence	4,48	0,56	4,46	0,61	0,827

 $\sigma$  (sigma) sd -Standard Deviation

As we can see in Table 8, students from all years of learning in nursing, have the highest average values in the Empathy dimension, but the highest between these groups are the  $3^{rd}$  (X=5,09; sd=0,87) and the  $1^{st}$  year (X=5,02;sd=0,85), and in contrast, it appears that the dimension with lower averages is Self-motivation, namely for the students in the  $1^{st}$  year (X=4,21; sd=0,77) and the  $2^{nd}$  year (x=4,29; sd=0,73),

and the Relationship Management presents also the lower overages in the  $2^{nd}$  (X=4,29; sd=0,76) and to the  $4^{th}$  (X=4,33;sd=0,75) years students.

Global Emotional Competence, with no statistically significant differences between groups (p= 0,599), presents in all academic years values up the cut point, but lower in  $2^{nd}$  (4,44; sd=0,52), and the highest is of  $4^{th}$  years students (X=4,58; sd=0,53).

Considering the values of standard deviations, the variables that represent each of the dimensions of Emotional Competence present a normal distribution. There are no statistically significant differences in the EVCE-r33 dimensions depending on the year the nursing student attends.

EC Dimensons	1 <sup>st</sup>		2 <sup>nd</sup>	2 <sup>nd</sup>		3 <sup>rd</sup>		4 <sup>th</sup>	
	х	σ	х	σ	х	σ	х	σ	р
Self-awareness	4,42	0 <i>,</i> 97	4,32	0,85	4,36	0,96	4,74	0,74	0,198
Emotional Mangt	4,46	0,86	4,47	0,66	4,62	0,77	4,56	0,66	0,576
Self-motivation	4,21	0,77	4,29	0,73	4,28	0,66	4,36	0,57	0,784
Empathy	5,02	0,85	4,98	0,74	5,09	0,87	4,98	0,84	0,859
Relationship Mangt	4,30	0,82	4,29	0,76	4,43	0,83	4,33	0,75	0,735
Global EC	4,45	0,64	4,44	0,52	4,52	0,57	4,58	0,53	0,599

**Table 8** – Values of Mean (X) and stand deviation ( $\delta$ )relative the five EVCE-r33 dimensions depending on the sample academic year attended.

 $\sigma$  (sigma) sd -Standard Deviation; X-mean;

In brief conclusion and with regard to statistical analysis, it appears that the variables under study present a normal distribution, and that most of them do not present significant statistical differentiation power, in relation to the values of the profiles found.

The five dimensions of EVCEr33 present a Moderate Profile, as well as Global Emotional Competence.

#### Each of these dimensions can be learned, reinforced

There are two very brief metacognitive exercises in the conceptual-labour field of health care that need to be addressed, even if briefly. The first is that it is not true that the only emotionally challenging contexts are those in which clinicians deal with life-or-death decisions. Especially because not even in these they are completely alone. The second is that it is urgent to deconstruct the idea that classical training is enough. More than this, it is important to add that it is absolutely healthy and developmental, for any professional field, to expand the respective areas of exercise and training, and that this is precisely an expressive window of the state of the art of caregivers in general and nursing development in particular. And previous training completed cannot, and should never, be sufficient for the human fabric of any profession. If it is prior to the now, it does justice to and concerns an eventually current and emerging reality, but which is never static or immutable. And not just for professionals... Users, patients, and even family members, are currently consuming more culture and information, which is why, in their status as citizens, they learn to discover rights, and in these, the right to quality. And so, they tend to progressively demand more and better quality of care.

They prioritize details, dominate expressive languages of this quality, adding to the technical instrumental executory component, the attentional and relational component to feel therapeutic.

And studies have often shown that health professionals in general, and nurses in particular, without training or with insufficient training in emotional competence and intra and interpersonal communication skills, - namely in the expression of specific competences in the area of nursing related to a person in a critical situation, (38, 40, 44) – experience greater and more evident levels of stress and emotional exhaustion when trying to control their own emotions and those of others (47, 48,49, 40, 51), in different Nursing contexts, from the pre-hospital area to intensive care units.

It remains to be said, and as already presented above, that the criteria units related to the specific competences are explicit with regard to the interactions that build and promote the therapeutic relationship, and like all activities/executive actions, from decisions to the implementation of plans in stressful environments, require good levels of emotional self-awareness and emotional management, not least self-motivation. And considering the profiles already studied in nursing, their predictive power for Emotional Competence in nurses was verified (37).

However, there is still a lot to study and understand about how we can reinforce the skills and dimensions that constitute Emotional Competence. It is essential to continue producing specific training in Emotional Education to develop all human beings in general in this type of soft skills, considered essential for relational interactions in the professional fabric of health contexts.

#### REFERENCES

- 1. Pérez-González, J. C., Yáñez, S., Ortega-Navas, M. C. y Piqueras, J. A. (2020). Educación emocional en la educación para la salud: cuestión de salud pública. *Clínica y Salud*, 31(3), 127-136. <u>https://doi.org/10.5093/clysa2020a7</u>
- Almeida, J., Vilela, C. & Veiga-Branco, M A.R. (2021). Competência Emocional dos enfermeiros no contexto dos Cuidados de Saúde Primários [Apresentação de póster]. Reunião Científica do Grupo NursID 21 Desafios e boas práticas na investigação em enfermagem, Porto, Portugal. <a href="http://nursid.esenf.pt/index.php/reuniao-cientifica-nursid/">http://nursid.esenf.pt/index.php/reuniao-cientifica-nursid/</a>
- Carício, M.R., Sousa MF, Alvarenga, J.P., Costa, L.D., Leandro, S.S., Jesus, E.A., Mendonça, A.V.M. (2021). A educação emocional pode influenciar na qualidade das competências e habilidades da Enfermagem?. *Enferm Foco*. 2021;12(Supl.1):15-21. DOI: 10.21675/2357-707X.2021.v12.n 7Supl.1.5212
- 4. Donabedian, A. (2003) An Introduction to Quality Assurance in Health Care. *Oxford University Press*, Oxford.
- Stanojlović, O., Šutulović, N., Hrnčić, D., Mladenović, D., Rašić-Marković, A., Radunović, N., & Vesković, M. (2021). Neural pathways underlying the interplay between emotional experience and behaviour, from old theories to modern insight. Archives of Biological Sciences, 73(3), 361–370. <u>https://doi.org/10.2298/ABS210510029S</u>
- 6. Damásio, A. (2000). *O Mistério da Consciência*. *Do corpo e das emoções ao conhecimento de si*. São Paulo. Companhia das Letras. Ed. Schwarcz Lda.
- 7. Myruski, S. & Dennis-Tiwary, T. A. (2022). Observed parental spontaneous scaffolding predicts neurocognitive signatures of child emotion regulation, *International Journal of Psychophysiology*, Volume 177, Pages 111-121,
- https://doi.org/10.1016/j.ijpsycho.2022.05.004.
- 8. Gross (2015). Emotion regulation: current status and future prospects, *Psychol. Inquiry* 26(1), 1–26. <u>http://doi.org/10.1254/fpj.151.21</u>
- 9. Decety & Jackson, (2004). The Functional Architecture of Human Empathy. Behavioural and Cognitive Neuroscience Reviews, Volume 3 Nr 2, Pages 71-100, https://doi.org/10.1177/1534582304267187
- 10. Langeslag, J.E. and Surti, K. (2017). The effect of arousal on regulation of negative emotions using cognitive reappraisal: An ERP study. *International Journal of Psychophysiology* 118 (2017) 18–26. http://dx.doi.org/10.1016/j.ijpsycho.2017.05.012
- 11. Ochsner, K. N, Gross, J. J. (2008). Cognitive emotion regulation: insights from social cognitive and affective neuroscience. *Curr. Direct. Psychol. Sci.* 17(2), 153-158. <u>http://doi.org/10.1111/j.1467-8721.2008.00566.x</u>
- 12. Damásio, A. (2010). *O Livro da Consciência*. *A Construção do cérebro consciente*. Temas e Debates.
- 13. Davis, M. H. (1983). Measuring individual differences in empathy: Evidence for a multidimensional approach. *Journal of Personality and Social Psychology*, 44(1), 113–126. https://doi.org/10.1037/0022-3514.44.1.113
- 14. Ickes, W. J. (Ed.). (1997). *Empathic accuracy*. The Guilford Press.

- 15. Decety, J., & Jackson, P. L. (2004). The functional architecture of human empathy. *Behavioural and cognitive neuroscience reviews*, 3(2), 71–100. <u>https://doi.org/10.1177/1534582304267187</u>
- 16. Hoffman, M. (2000). Implications for Socialization and Moral Education. In Empathy and Moral Development: Implications for Caring and Justice (pp. 287-298). Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511805851.013</u>
- 17. Reeck C. and Onuklu N.N.Y. (2022) Interpersonal Emotion Regulation: Consequences for Brands in Customer Service Interactions. Front. Psychol. 13:872670. http://doi: 10.3389/fpsyg.2022.872670
- Niven, K., Macdonald, I. and Holman, D. (2012). You spin me right round: cross-relationship variability interpersonal emotion regulation. *Front. Psychol.*, Sec. Emotion Science, Vol 3 <u>https://doi.org/10.3389/fpsyg.2012.00394</u>
- Bloch, L., Haase, C. M., & Levenson, R. W. (2014). Emotion regulation predicts marital satisfaction: more than a wives' tale. Emotion (Washington, D.C.), 14(1), 130–144. <u>https://doi.org/10.1037/a0034272</u> T
- 20. Roberton, T., Daffern, M. & Bucks R. S. (2014) Maladaptive emotion regulation and aggression in adult offenders, Psychology, Crime & Law, 20:10, 933-954, DOI: 10.1080/1068316X.2014.893333
- 21. Davis, S. K., and Qualter, P. (2020). Emotional competence. *The Encyclopedia of Child and Adolescent Development*, eds S. Hupp and J. Jewell (Wiley). https://doi: 10.1002/9781119171492.wecad
- 22. Saarni, C. (1999). *The Development of Emotional Competence*. Guilford Pres.
- 23. Saarni, C. (2000). Emotional Competence: The social Construction of the Concept. In R. Bar-on & J. Parker. (Eds.). *The Handbook of Emotional Intelligence*. Jossey-Bass.
- 24. Bisquerra, R. (2003). Educación emocional y competencias básicas para la vida. *Revista de Investigación Educativa* (RIE), 21 (1), 7-4.
- 25. Bisquerra, R., Perez, N. (2007). Las competencias emocionales. Educación XXI: *Revista de la Facultad de Educación*, (10), 61-82.
- 26. Fragoso-Luzuriaga, R. (2015). Inteligencia emocional y competencias emocionales en educación superior, ¿un mismo concepto? Revista Iberoamericana de Educación Superior, 6, 16, 110-125
- 27. Veiga-Branco, M.A.R. (2004). Competência emocional. Quarteto. http://hdl.handle.net/10198/5463

28. Veiga-Branco, M.A.R. (2005). Competência Emocional em Professores. Um discurso do campo educativo. Tese Doutoramento. *Faculdade de Psicologia e Ciências da Educação da Universidade do Porto*. (FPCEUP). http://hdl.handle.net/10198/5311

- 29. Goleman, D. (1995). Emotional intelligence. Bantam Books, Inc.
- Veiga-Branco, M.A.R. (2007). Competência Emocional em Professores. In (Coord.) Candeias, A. A., Almeida, L. S. (2007). *Inteligência Humana. Investigação e aplicações*. Quarteto. 361-379. http://hdl.handle.net/10198/5278

31. Veiga-Branco, M.A.R. (2009). Escala Veiga Branco das capacidades da inteligência emocional (EVBCIE): partes I e III. In Mendonça, S. S. (2009). *Competências Profissionais dos Enfermeiros: a Excelência do Cuidar*. Novembro. 162-164. http://hdl.handle.net/10198/5283

32. Goleman, D. (1999). Trabalhar com Inteligência Emocional. Temas e Debates.

- 33. Veiga-Branco, M.A.R., Antão, C., Ribeiro, I., Vaz, j., Ferro-Lebres, V. (2011). Emotional competence profile in a higher education students sample. In III International Congress of Emotional Intelligence: book of abstracts. Opatija <u>http://hdl.handle.net/10198/6916</u>
- Veiga-Branco, M.A.R., Afonso, J., Caetano, I. (2012). Emotional competence in healthcare professionals: correlational and comparative research. in IEA-EEF European Congress of Epidemiology 2012: Epidemiology for a Fair and Healthy Society. Porto. s57-s57 <u>http://hdl.handle.net/10198/8392</u>
- 35. Afonso, J,. Fernandes, A. & Veiga-Branco, M.A.R. (2013). Competência emocional (CE) nos fisioterapeutas da rede nacional de cuidados continuados integrados (RNCCI). In VI Jornadas de Tecnologias da Saúde. Guarda
- Veiga-Branco, M.A.R. & Costa, V. (2015). As competências emocionais nos decisores políticos portugueses. In (Ed.) Celeste Simões, C., Evans, K., e Lebre, P. (2015). *Book of Abstracts. 5th ENSEC Conference. Social Emotional learning and Culture*. Faculdade de Motricidade Humana. Lisboa. P. 78.
- 37. Almeida, J., Vilela, C. & Veiga-Branco, A. (2021). Competência Emocional dos enfermeiros no contexto dos Cuidados de Saúde Primários [Apresentação de póster]. *Reunião Científica do Grupo NursID 21 Desafios e boas práticas na investigação em enfermagem*, Porto, Portugal. <a href="http://nursid.esenf.pt/index.php/reuniao-cientifica-nursid/">http://nursid.esenf.pt/index.php/reuniao-cientifica-nursid/</a>
- 38. Moreira, D., Laranjeira, C., Gomes, L.G. (2021). A Competência Emocional do Enfermeiro e a Comunicação terapêutica face à pessoa com manifestações de perturbação mental. Mestrado em Enfermagem de Saúde Mental. Instituto Politécnico de Leiria, IPL - Dissertações de Mestrado. Leiria. In press.
- 39. Nogueira, S., Veiga-Branco, M.A.R. & Monte, A.P. (2021). Competência Emocional em decisores autárquicos. Comunicação em Seminário: Inteligência Emocional – da visão científica à aplicação prática. 1ª Jornadas da Associação Portuguesa de Inteligência Emocional (APIE). 6 novembro, 2021. Evento on-line.
- 40. Rodrigues, P.A.R. (2017). Determinantes da competência emocional de profissionais em saúde na abordagem ao doente crítico. Mestrado em Enfermagem Médico-Cirúrgica. *ESSa Dissertações de Mestrado Alunos. IPB.* <u>http://hdl.handle.net/10198/14784.</u>
- 41. Ferreira, C. (2018). Competência Emocional e Interdisciplinaridade nas Equipas Referenciadoras das Unidades de Cuidados de Saúde Primários: Uma Perspetiva de Cuidadores Formais e Informais. Mestrado em Cuidados Continuados ESSa - Dissertações de Mestrado Alunos. IPB
- 42. Giehl, B., Monte, A.P., Veiga-Branco, M.A.R. (2020). *Relationship between investor's profile and the level of emotional competence*. Contabilidade e Finanças. ESTiG.D Dissertações de Mestrado Alunos . <u>http://hdl.handle.net/10198/23233</u>

- 43. Silva, A.M.G., Monte, A.P., Veiga-Branco, M.A.R. (2023). Perfil dos Empresários do Distrito de Bragança e Competência Emocional – estudo correlacional. Contabilidade e Finanças. ESTiG.D Dissertações de Mestrado Alunos.
- 44. Veiga-Branco, M.A.R., Lopes, T. (2012). Nursing emotional competence profile: exploratory study in continued care national network. In The IEA-EEF European Congress of Epidemiology 2012: *Epidemiology for a Fair and Healthy Society*. s82-s82 <u>http://hdl.handle</u>.net/10198/8390
- 45. Agostinho, L. M. C. F. (2008). Competência Emocional em Enfermeiros: Um estudo em hospitais públicos. *Universidade de Aveiro* (Portugal).
- 46. Hair, J.F., Black, W.C., Babin, B.J. and Anderson, R.E. (2010) *Multivariate Data Analysis*. 7th Edition, Pearson, New York.
- 47. Paiva, J. D. M., Cordeiro, J. J., Silva, K. K. M. D., Azevedo, G. S. D., Bastos, R. A. A., Bezerra, C. M. B., ...
  & Martino, M. M. F. D. (2019). Fatores desencadeantes da síndrome de burnout em enfermeiros. *Rev. Enferm.* UFPE on line, 483-490.
- 48. Martins P., S., Fonseca, A. M., & Carvalho, A.S. (2011). Burnout in palliative care: A systematic review. *Nursing Ethics*, 18(3), 317–326. <u>https://doi.org/10.1177/0969733011398092</u>
- 49. Nobre, D.F.R., Rabiais, I.C.M., Ribeiro, P.C.P.S.V., Seabra, P.R.C. (2019). Burnout assessment in nurses from a general emergency service. *Rev Bras Enferm*. 2019;72(6):1457-63. doi: http://dx.doi.org/10.1590/0034-7167-2017-0870
- Sé, A. C. S., Machado, W. C. A., Passos, J. P., da Silva Gonçalves, R. C., Cruz, V. V., Bittencourt, L. P., Paiva, A.P.D.L. & de Figueiredo, N. M. A. (2020). Prevalência da síndrome de burnout em enfermeiros do atendimento pré-hospitalar. Research, Society and Development, 9(7), e940975265-e940975265.
- 51. Souza, J.D., Pessoa J.J.M., Miranda F.A.N. (2017). Stress in an emergency room and the challenges for Brazilian and Portuguese nurses. Rev Enferm Ref. IV(12):107–16. doi: <u>https://doi.org/10.12707</u> /<u>RIV16064</u>

## CHAPTER 7 Virtual Reality in Emotional Competence Education

Authors: Maria CASSAR, Daren CHIRCOP, Roderick BUGEJA

#### 7.1 Virtual Reality (VR)

With the modernisation and advancement of healthcare and clinical environments, and simultaneously a rapid growth and evolution in computer science and technology, so has the call and demand for safe and efficient clinical nurse training grew over the years. Growing opportunities exist to modernize healthcare education into programs designed to satisfy the expectations and needs of the digital generation, due to technological breakthroughs and their affordability (King et al., 2018). As a result, Virtual Reality (VR) has been becoming increasingly popular integrated into professional nursing curricula with the promise of revolutionizing nursing education, streamlining teaching and learning procedures, offering a fresh and interesting approach to nurse training (Park, Kim and Kim, 2020). Indeed, VR can be used as a pedagogical resource for all health care professionals' training (Ferrandini Price et al., 2018).

VR generates 4D visual and auditory scenes that can be navigated and interacted with (Vince, 2004). Although it is unclear when virtual reality technology first entered nursing education, simulation and computer-based technologies have been used and researched as a teaching tool by Ivan Sutherland since the late 1960s with the term VR believed to have been first coined by Jaron Lanier in 1987 (Arvind Mallik, 2018). Utilizing a headset that covers the user's eyes and displays the virtual environment, together with hand-held controllers or other input devices to interact with the environment, a user is often fully immersed in a VR experience (Bailenson & Yee, 2007), eliminating a real-world interaction (Azuma, 1997 ; Mills et al., 2022; Vince, 2004). With such tool, users can interact with digital objects and settings as if they were real through a computer-generated simulation (Riva, Wiederhold, & Wiederhold, 2016). VR technologies enable a new perceptual extension, creative construction and a unique social involvement (Bricken & Byrne, 1993).

#### 7.2 Virtual reality in Nursing Education

Virtual reality (VR) has been employed to nursing education, improving teaching and learning methods, thus aiding in nursing students' skill development, confidence building and clinical practice preparation (Chen et al., 2020; Biyik Bayram & Caliskan, 2022). Additionally, VR technology has been shown to improve students' critical thinking and problem-solving abilities as they get to practice and apply their knowledge in a secure and regulated setting (Lang et al., 2018). VR technology offers nursing students the chance to practice and explore a multitude of skills, hence learn by doing, in a safe, secure and controlled setting without putting patients at risk or harm, and hone their clinical reasoning abilities in a more productive and enjoyable learning environment (Alimohammadi, Norouzi, & Alhani, 2017). In fact, evidence-based findings has shown that VR technology in tertiary education offers nursing students a unique and engaging learning environment that can boost motivation, improving learning results (Kim & Kim, 2018).

Furthermore, Bender et al. (2018) has shown that utilizing virtual reality simulations to practice patient care scenarios enhanced patient care abilities and skills more effectively, and with less fear and anxiety than when caring for real patients. As a result, VR in nursing education has also benefitted patient safety (Biyik Bayram & Caliskan, 2022), enhancing patient care outcomes (Pfeiffer et al., 2019) and patient satisfaction (Gee et al., 2020). While much published scientific evidence has shown that VR technology has been employed in various nursing education programs to address a wide range of skills and topics, no evidence was encountered that prior to this project VR technology has been used to help develop emotional competence skills.

Notwithstanding the above, nurse educators must not consider VR technology as a fool-proof aid to nursing education. Albeit the benefits of integrating an immersive technology to nurse training, one must be cautioned against other realities which may entail a variety of risks or other challenges necessitating preventing actions. These include:-

- i. **Physical discomfort**: Some students may experience physical pain or headaches from VR simulations due to prolonged use or an uncomfortable fit of the VR gear, known as cybersickness (Cobb et al., 1999; Davis et al., 2015; Obukhov et al., 2022).
- Physiological distress: In some people, especially those who are unprepared or untrained in its use, VR simulations can also lead to worry, stress, or even panic in certain situations (Costello, 1997). Some students need more time to understand and adjust to it (Chang & Lai, 2021).
- iii. **Health and safety concerns**: Since VR technology immerses the user in a simulated environment, there is a possibility of danger or injury if the user becomes unaware of their immediate surroundings. Use of VR necessitates large dedicated spaces with safety measures in place.
- iv. **Cost**: One of the primary challenges to applying VR in nursing education is the cost of equipment, its maintenance, and the cost of developing and updating VR simulations which can be very high, making it difficult for some nursing programmes to justify the investment.
- v. **Technical issues**: Due to its complexity and tendency for technical issues, VR technology has the potential to disrupt learning processes and waste time and resources.
- vi. Lack of Institutional support: several public forum participants noted that getting faculty support was a recurrent problem. Faculty members frequently lack knowledge of, or inadequate grasp of the technology, so participants remarked that this challenge must be overcome by providing evidence of VR's usefulness (Cook et al., 2019).
- vii. Desensitization: While VR can offer a safe and controlled environment for students to practice their skills, it may not fully replicate the complexity of real-life patient situations. There is a risk of desensitization to real-life patient situations. Students may become desensitized to certain situations or may not fully develop the interpersonal and communication skills necessary for patient care. This risk highlights the importance of using VR simulations in conjunction with clinical practice or simulation labs, thus supplemental to traditional means of training rather than replacing them, because human interaction is crucial especially in learning social and communication skills (van der Lindevan den Bor et al., 2022).

#### 7.3 The SG4NS Serious Game

The gaming industry produces VR games which allow for the study of the users' experience and emotions while playing the game. One of the best game genres that allow for this type of experience is known as Escape Rooms (Oliveira, 2022). The latter are a genre of game focused on problem-solving involving puzzles and mysteries that can be solved through riddles and symbols, challenging the player psychologically and allowing a relationship between the player's mind and body (Oliveira, 2023).

The development of such game requires a thorough understanding of how emotions are aroused in the human body, and in turn, the game becomes very effective at eliciting emotions in the player, involving the latter more deeply in the experience, allowing for the best possible gaming experience. The use of virtual reality technology further allows 'full immersion', with a sense of 'presence', 'feeling in the body' as if the experience was real, leading to greater player involvement and active participation in the experience (Oliveira, 2022).

The central goal of incorporating a Serious Game into the Emotional Competence Course was to invoke a spectrum of emotions in students within a controlled and captivating environment, while delivering an immersive and engaging experience to further motivate students' active participation. The latter Escape Room game genre presented this project with the ideal scenario for a VR Serious Game tool to assist the teaching and learning of emotional competence skills to the students. The potential of VR technology and the Escape Room genre as a powerful means of eliciting emotions in players has been demonstrated in a study by Oliveira et al. (2023).

To ensure accessibility for all students, the game was intentionally designed with a simple interface and minimal controls. This approach catered to students unfamiliar with virtual reality, allowing them to navigate and engage with the game effortlessly. The emphasis on simplicity aimed to remove potential barriers, enabling a broader range of students to benefit from the emotional exploration facilitated by the Serious Game within the course.

The game commences with the student/gamer finding oneself in a house, exactly within a kitchen environment, in a dark and unlit environment. Some parts of the house were completely dark and has to be navigated with a flashlight. Written post-it notes on the kitchen-top guide the student to further clues to explore the home environment which consists of a number of rooms, including a living room, a bedroom, a balcony, a box room, an office and a laboratory found on two floors, unravelling other puzzles during the process. The overall ambience is unlit, ghostly (by way of strange sounds and whispers from a woman along the stairway), designed in a manner to stimulate the student's emotions. Among the instructions given, the gamer must open a trash bin using a hammer, requiring a tough action to retrieve a key from within the trash surrounded by cockroaches. The students must navigate around the house, opening doors and following instructions to gather a number of keys and pages from a diary, which are dated and followed by text signed "John".

At one point, the gamer is faced with a seemingly unconscious, albeit not dead woman lying on the floor, who is "Mary". Mary's body changes position every time the player looks away. When the player approaches the body, Mary's eyes follow the player. At another point, the player has to pass through a hole in the balcony and walk over a narrow plank, eliciting a fear of height. The narrative of the game focuses on the lives of the game's characters, John and Mary. Each six pages of the diary features a date which the player needs to connect in a specific order to obtain a code, which opens a cabinet in the laboratory. Upon successfully opening it, the player sees their reflection in the mirror and realises they are "John". The author of the game claims that although there is this storyline, it remains open-ended, allowing each player to have their own personal interpretation of what happened. The objective of the game is to convey the narrative to the player in a way that is engaging and arouses curiosity about what happened (Oliveira et al., 2023).

Following gameplay, students actively participated in open discussions during the course, recalling and analysing the emotions they experienced while navigating the virtual environment. Among the array of positive and negative emotions experienced, the students highlighted fear, apprehension, disgust, curiosity, mystery, concern, helplessness, frustration, boredom, confusion, anger, surprise, shock/alarm, excitement, anxiety, tension, embarrassment amazement, wonder, awe and satisfaction. The structured

post-game discussions provided a platform for students to articulate and reflect on their emotional responses, fostering a deeper understanding of their own emotional competencies. This reflective component was instrumental in linking the virtual experience to real-world emotional competence, enriching the overall learning process.

#### References

- Alimohammadi, M., Norouzi, R., & Alhani, F. (2017). The use of virtual reality in nursing education: A systematic review. *Nurse Education Today*, 44, 33-38. https://doi.org/10.1016/j.nedt.2016.06.019
- Arvind Mallik, D. M. (2018). Virtual Reality in Higher Education- Enhance Teaching Delivery by leveraging Quality Learning. *International Journal of Research and Analytical Reviews*, 5(4), 947– 950.
- Azuma, R. T. (1997). A survey of augmented reality. *Presence : teleoperators and virtual environment*, 6(4), 355–385. 10.1162/pres.1997.6.4.355
- Bailenson, J. N., & Yee, N. (2007). The ecology of avatar presence: Psychological, social, and cultural aspects of a new form of social relationship. *Presence: Teleoperators and Virtual Environments*, 16(4), 305-329. <u>https://doi.org/10.1111/j.1468-2958.2007.00299.x</u>
- Bender, K., Kilinc, E., Yang, J., & Kim, J. H. (2018). The effects of virtual reality simulation on nursing students' communication skills and anxiety. *Journal of Nursing Education and Practice*, 8(10), 58-64. <u>https://doi.org/10.5430/jnep.v8n10p58</u>
- Biyik Bayram, S., & Caliskan, N. (2022). The use of virtual reality simulations in nursing education, and Patient Safety. Contemporary Topics in Patient Safety - Volume 1. <u>https://doi.org/10.5772/intechopen.94108</u>
- Bricken, M., & Byrne, C. M. (1993). Summer students in virtual reality. *Virtual Reality*, 199–217. <u>https://doi.org/10.1016/b978-0-12-745045-2.50019-2</u>
- Chang, Y. M., & Lai, C. L. (2021). Exploring the experiences of nursing students in using immersive virtual reality to learn nursing skills. *Nurse Education Today*, *97*, 104670. <u>https://doi.org/10.1016/j.nedt.2020.104670</u>

- Chen, F.-Q., Leng, Y.-F., Ge, J.-F., Wang, D.-W., Li, C., Chen, B., & Sun, Z.-L. (2020). Effectiveness of virtual reality in nursing education: Meta-analysis. *Journal of Medical Internet Research*, 22(9). <u>https://doi.org/10.2196/18290</u>
- Cobb, S. V., Nichols, S., Ramsey, A., & Wilson, J. R. (1999). Virtual reality-induced symptoms and effects (VRISE). *Presence: Teleoperators and Virtual Environments*, 8(2), 169–186. <u>https://doi.org/10.1162/105474699566152</u>
- Cook, M., Lischer-Katz, Z., Hall, N., Hardesty, J., Johnson, J., McDonald, R., & Carlisle, T. (2019). Challenges and strategies for Educational Virtual reality. *Information Technology and Libraries*, *38*(4), 25–48. <u>https://doi.org/10.6017/ital.v38i4.11075</u>
- Costello, P.J. *Health And Safety Issues Associated with Virtual Reality: A Review of Current Literature;* Advisory Group on Computer Graphics: Loughborough, UK, 1997
- Davis, S., Nesbitt, K., & Nalivaiko, E. (2015, January). *Comparing the onset of cybersickness using the Oculus Rift and two virtual roller coasters*. In Proceedings of the 11th Australasian conference on interactive entertainment (IE 2015) (Vol. 27, p. 30). Sydney, Australia: Australian Computing Society.
- Faiola, A., & Bognar, A. (2019). Simulation in nursing education: A review of the research. *Journal of nursing education and practice*, 9(3), 26-33.
- Ferrandini Price, M., Escribano Tortosa, D., Nieto Fernandez-Pacheco, A., Perez Alonso, N., Cerón Madrigal, J. J., Melendreras-Ruiz, R., García-Collado, Á. J., Pardo Rios, M., & Juguera Rodriguez, L. (2018). Comparative study of a simulated incident with multiple victims and immersive virtual reality. *Nurse Education Today*, *71*, 48–53. https://doi.org/10.1016/j.nedt.2018.09.006
- Gee, J. P., Hardy, K. K., & Cedel, L. L. (2020). The impact of virtual reality on patient satisfaction and health outcomes. *Journal of Nursing Education and Practice*, 10(2), 15-22. <u>https://doi.org/10.5430/jnep.v10n2p15</u>
- Kim, C. H., & Kim, H. J. (2018). Virtual reality in nursing education: A review of the literature. Journal of Educational Technology Development and Exchange, 11(1), 1-12. <u>https://doi.org/10.17783/etde.2018.11.1.1-12</u>
- King, D., Tee, S., Falconer, L., Angell, C., Holley, D., & Mills, A. (2018). Virtual health education: Scaling practice to transform student learning. *Nurse Education Today*, *71*, 7–9. <u>https://doi.org/10.1016/j.nedt.2018.08.002</u>

- Lang, E., Heger, M., & Bauernfeind, T. (2018). Virtual reality in nursing education: A review of the literature. *Journal of Nursing Education and Practice*, 8(6), 46-53. <u>https://doi.org/10.5430/jnep.v8n6p46</u>
- Park, Y. J., Kim, E. J., & Kim, J. H. (2020). The effects of virtual reality-based nursing education on learning outcomes and attitudes toward nursing education. *Journal of Clinical Nursing*, 29(19-20), 2907-2917. <u>https://doi.org/10.1111/jocn.15147</u>
- Mills, K. A., Scholes, L., & Brown, A. (2022). Virtual Reality and Embodiment in Multimodal Meaning Making. *Written Communication*, 39(3), 335-369. <u>https://doi.org/10.1177/07410883221083517</u>
- Obukhov, A. D., Krasnyanskiy, M. N., Dedov, D. L., & Nazarova, A. O. (2022). The study of virtual reality influence on the process of professional training of miners. Virtual Reality : *The Journal of the Virtual Reality Society*, 1-25. <u>https://doi.org/10.1007/s10055-022-00687</u>
- Oliveira, I., Oliveira, E. and Gomes, L. (2022). Escape VR: the guilt : um escape room em realidade virtual para elicitação de emoções. Projeto apresentado ao IPCA, para obtenção do grau de Mestre em Engenharia de Desenvolvimento de Jogos Digitais. April, 2022.
- Oliveira, I., Carvalho, V., Soares, F., Novais, P., Oliveira, E., and Gomes, L. (2023) Development of a Virtual Reality Escape Room Game for Emotion Elicitation. *Information* 2023, 14, 514. https://doi.org/10.3390/ info14090514
- Pfeiffer, E., Smith, K., & Quirk, R. (2019). The impact of virtual reality on anatomy and physiology education in nursing students. *Anatomical Sciences Education*, 12(1), 18-26. <u>https://doi.org/10.1002/ase.1784</u>
- Riva, G., Wiederhold, B. K., & Wiederhold, M. D. (2016). Virtual reality in clinical psychology and neuroscience. *Frontiers in Psychology*, 7, 708. <u>https://doi.org/10.3389/fpsyg.2016.00708</u>
- van der Linde-van den Bor, M., Frans-Rensen, S. A., Slond, F., Liesdek, O. C. D., de Heer, L. M., Suyker, W. J. L., Jaarsma, T., & Weldam, S. W. M. (2022). Patients' voices in the development of pre-surgical patient education using virtual reality: A qualitative study. *PEC Innovation*, *1*, 100015. <u>https://doi.org/10.1016/j.pecinn.2021.100015</u>

Vince, J. (2004). Introduction to virtual reality. Springer Science & Business Media.

### CHAPTER 8 Conclusion and Future Recommendations

Authors: Maria CASSAR, Daren CHIRCOP, Roderick BUGEJA

#### 8.1 Summary

Changes in health care systems, the nursing profession and nursing practice have elicited new demands for nurse education and nurse educators. The nurse workforce, nevertheless, remains the largest population of health care workers in most countries. On average, across the EU counties, there are 8.3 nurses per 1 000 population in 2020 in EU countries (The Organisation for Economic Co-operation and Development [OECD]/European Union, 2022). Increasing prevalence of chronic diseases, ageing population, global health issues (Shustack, 2020), e.g. pandemics (Lake, 2020), many environmental risks (Alvarez-Nieto et al., 2018) and technological solutions have enabled modifications and shifts in changed health care systems. The nursing profession and nursing practice have sought to adapt accordingly (European Commission, 2017). Nurses remain central to the efficient and effective delivery of care in hospital settings, long-term care facilities, and the primary care and community settings. The competence of nurses therefore maintains crucial importance in health care systems. The quality of nursing care and the education of nursing students remains the fulcrum of good health systems.

Against this backdrop, the demands and the needs to adopt new teaching approaches and methods, and adaptive strategies to nurse educators' work and competence are indicated (Salminen et al., 2021). It is hoped that this handbook contributes to the narratives and efforts seeking to optimize nurse educators' work and competence. The competence of nurse educators is essential if high quality and evidence-based nurse education is to be offered towards developing student nurses' knowledge, skills and attitudes (World Health Organization [WHO], 2021). The authors of this handbook trust that this handbook is a valid and useful contribution towards the development of nurse educators' required competence in enabling and guiding the development of emotional intelligence amongst nurse students.

In essence, this handbook is an eclectic resource of information and reflections for nurse educators regarding the development of emotional intelligence amongst nursing students. The development of emotional intelligence amongst nursing students has been distinguished as a central asset of nurse education. This contention echoes through the project and through all its deliverables and outputs including this handbook. Specifically, this chapter provides recommendations for nurse educators for the future, in this regard. In the following sections academics from the SG4NS project share the outcomes of reflection and evaluation exercises which they pursued in the final months of the project period. The recommendations draw upon a critical analysis of their experience in the project activities and upon the evaluation feedback which they gathered in the course of conducting the project between 2020-2023 across six project partner universities in Portugal, Italy, Romania, Spain and Malta.

The SG4NS project (2020-2023) was designed and conducted to address the demands in modern health care systems referred to above and to contribute towards the development of the nurse educator practice with regards to their efforts towards developing emotional intelligence amongst nursing students. With this project the first steps to address the need for novel and effective approaches to developing emotional intelligence were paced.

The project comprised several activities and outputs including the conduct of a literature review, the development of a Serious Game which implied the use virtual reality immersive technology and a instructional workshop; all with a focus on developing emotional intelligence amongst nursing students. The instructional workshop was well received by the attendees who volunteered participation. The multinationalism of the participants was deemed an enriching facet of the project. The recommendations below emanate from all of the above.

# 8.2 Future recommendations for nurse educators' contribution to development of emotional competence amongst nursing students

#### Recommendation 1: Universal nursing students' "emotional intelligence development" requirements

The requirements of nursing students' emotional intelligence need to be discussed and harmonised at a pan-European level, and gradually and eventually at a global level. In view of the phenomena of globalisation and nurse mobility across the global nurse workforce, universal consensus regarding the requirements of "emotional intelligence development" of nursing students is indicated. Further research is suggested to seek guidance in designing "emotional intelligence development" strategies which are universal enabling the provision of and access to educational resources which can guide and support the development of emotional intelligence amongst students across the globe. The SG4NS project comprised an initial step in this direction. Wastage and inefficiency, arising from fragmentation of investments and duplication of initiatives, are challenges which nurse educators are currently experiencing (Leaver et al, 2022), hence, international collaborations are viewed as real sustainable solutions to address identified gaps and discrepancies in nurse education regarding emotional intelligence development. The determination of universal nursing students' emotional intelligence requirements will consolidate and economise expertise, resources, initiatives and investments across different countries and is therefore recommended.

# Recommendation 2: Develop international nurse educators' educational opportunities, regarding the development of emotional intelligence amongst nursing students

There is a need to model nurse educator competence and to base nurse educator education on universal guidelines and parameters which draw, in turn, on the best available evidence. This recommendation is being proposed in view of advocating the benefits of universal levels of development of emotional intelligence amongst nursing students in current realities, referred to above. The literature extensively calls for improved and more equality across the quality of nursing education globally so as to address disparities in nursing service quality across the globe (Satoh et al.2020). Continuing professional development (CPD) is a self-directed learning process (Drude et al., 2019) that is called for throughout

the duration of an educator's activity. A universal structured CPD programme/s for nurse educators to support the "emotional intelligence development "of nursing students is recommended. This CPD will allow for continued development of nurse educators' competence in this regard (Pool et al., 2015) and will enable them to meet dynamic realities over their years of professional practice as educators.

# *Recommendation 3:* Direct nurse education institutions and nurse educators' employers to channel resources, including time and money, to support nurse educators' strategies and approaches which are directed towards the development of emotional intelligence of nursing students

Advances in the use of technology in education and changing characteristics of student populations require knowledge and skills which are associated with modern modes and methods of teaching and learning. On one hand global and work management challenges faced by nurse educators (EU-OSHA 2018, Howard et al., 2022) including technological advances alongside demographic changes across Europe that increase the demand on health care and a shortage of nurses and nurse educators, are recognized elements of current nurse education institutions. On the other hand, nurse educators suffer from high workloads which is typically unevenly distributed over the academic year (Rinne et al., 2022). In addition they experience mental strain and work-related stress (Singh, 2020) The resources required to develop knowledge and skills sets and the resources needed by nurse educators to integrate and use new technologies effectively in their teaching are to receive more attention and investment. Insights from this project support the view that there is little research on the needs of nurse educators in this regard. This knowledge gap highlights the need for future research. It is important to avoid overlooking the fact that individual educators and country specific contexts have different requirements because (1) initial starting lines vary across context and (2) the rate and nature of the dynamism in a specific context also varies from any other. Investment that is channelled specifically to emotional intelligence development is therefore recommended.

# Recommendation 4: Establish an international hub for "emotional intelligence development" for the nursing profession

The sharing of knowledge and expertise between the partner entities in the SG4NS has provided a rich space for the learning and development. Therefore the establishment of a hub for the "emotional intelligence development" for the nursing profession is recommended to advance the quality development of emotional intelligence across the globe. One of the main functions of the hub will be to foster collaborations between entities in different countries as well as to enable exchanges between individual educators in-person and of resources, experiences and expertise. Such a hub is believed to have the potential to increase the possibilities for professional international collaboration in a structured and systematic manner to enhance unified and harmonised education of nursing students across the globe in this regard. The learning and growth that arose from the transnational experience of this project is commendable and further learning and growth of this nature is recommended. The hub will allow for assimilation, review, and evaluation of future research studies in view of effective dissemination of evidence and best practice In the least, the recommended hub will serve as a repository for resources that facilitate teaching and learning for nurse educators. The outputs of the SG4NS project, including the serious game and the literature review and workshop will be the founding components of this hub. Other resources will be co-created by educators or teams of educators and shared across nations, to support the future's nurse educator workforce and the nursing students of the future. At most, the hub may (1) share advice regarding "emotional intelligence development "in the nurse workforce to global organisations as is the WHO and the ICN, and (2) will provide a formal working arena for research. This research activity is necessary as it provides the evidence that may inform policy, practice, teaching and learning.

Recommendation 5: The adoption of immersive technologies, including the use of serious games and VR headsets in enabling the development of emotional intelligence, requires caution and should be preceded by a situational analysis of the context

The SG4NS project utilised an immersive serious game including the use of VR headsets. Whilst proving to be an element of attraction for students to engage in a workshop regarding the development of emotional intelligence, gaps and limitations in the proficiency and comfort of both educators and students with the use of the referred technology tools and equipment presented significant challenges. Also, the duration of the learning activity which entailed the use of the VR headsets and the venue of the learning activity would have clearly benefitted from more wider considerations when planned. These challenges, upon evaluation, were identified as potential threats and inhibitors which jeopardised and compromised the teaching and learning to varying extents across different individuals.

In view of the above, it is recommended that prior to the conduct of an educational initiative utilising such tools and equipment as are VR headsets and serious games, the approach and strategy of the educational initiative is to be, cautiously mapped against an analysis of the situation, that is, of the context in which this educational initiative will be provided and delivered.

#### REFERENCES

Drude, K. P., Maheu, M., & Hilty, D. M. (2019). Continuing Professional Development: Reflections on a Lifelong Learning Process. *The Psychiatric clinics of North America*, 42(3), 447–461. <u>https://doi.org/10.1016/j.psc.2019.05.002</u>

Pool, I. A., Poell, R. F., Berings, M. G., & ten Cate, O. (2015). Strategies for continuing professional development among younger, middle-aged, and older nurses: a biographical approach. *International journal of nursing studies*, 52(5), 939–950. <u>https://doi.org/10.1016/j.ijnurstu.2015.02.004</u>

Howard A., Antczak R., & Albertsen K. 2022. Education – Evidence from the European Survey of Enterprises on New and Emerging Risks (ESENER). The European Agency for Safety and Health at Work (EU-OSHA). <u>https://www.bollettinoadapt.it/wp-</u>content/uploads/2022/10/Education%E2%80%93evidence-ESENER\_en.pdf

Rinne, J., Leino-Kilpi, H., Saaranen, T., Pasanen, M., & Salminen, L. (2022). Educators' occupational well-being in health and social care education. Occupational medicine (Oxford, England), 72(5), 289–297. https://doi.org/10.1093/occmed/kqac024

Singh, C., Cross, W, Munro, I., & Jackson, D. (2020). Occupational stress facing nurse academics—A mixed-methods systematic review. Journal of Clinical Nursing, 29(5-6), 720–735. https://doi.org/10.1111/jocn.15150

Leaver, C. A., Stanley, J. M., & Veenema, T. G. (2022). Impact of the COVID-19 pandemic on the future of nursing education. Academic Medicine, 97(3), S82.

Alvarez-Nieto, C., Richardson, J., Parra-Anguita, G., Linares-Abad, M., Huss, N., Grande-Gasc´on, L.,

Grose, J., Huynen, M., L'opez-Medina, I.M., 2018. Developing digital educational materials for nursing and sustainability

European Commission, 2017. State of health in the EU companion report 2017.

https://ec.europa.eu/health/sites/health/files/state/docs/2017\_companion\_en.pdf

Lake, M.A., 2020. What we know so far: COVID-19 current clinical knowledge and research. Clinical

Medicine (Lond) 20 (2), 124–127.

OECD/European Union (2022), Health at a Glance: Europe 2022: State of Health in the EU Cycle, OECD

Publishing, Paris, https://doi.org/10.1787/507433b0-en.

Shustack, L., 2020. Integrating Google Earth in community health nursing courses preparing globally aware nurses. Nurse Educ. 45 (2), E11–E12.

Satoh, M., Fujimura, A., & Sato, N. (2020). Competency of academic nurse educators. *SAGE Open Nursing*, 6, 2377960820969389. <u>https://doi.org/10.1177/2377960820969389</u>

7

Serious Games - Developing Emotional Competences for Nursing Students 2020-1-PT01-KA203-078847, co-financed by Erasmus + Program KA2 -Cooperation for Innovation and the Exchange of Good Practices, KA203 -Strategic Partnerships for higher education